brought to you by **CORE** 

ISSN 1825-0211



# FOREIGN INVESTMENTS AND PRODUCTIVITY EVIDENCE FROM EUROPEAN REGIONS

Davide CASTELLANI - Fabio PIERI

Quaderno n. 83 — Gennaio 2011

# QUADERNI DEL DIPARTIMENTO DI ECONOMIA, FINANZA E STATISTICA

# Foreign Investments and Productivity Evidence from European Regions $\stackrel{\Leftrightarrow}{\approx}$

Davide Castellani<sup>a,b</sup>, Fabio Pieri<sup>a,c</sup>

<sup>a</sup>Department of Economics, Finance and Statistics, University of Perugia <sup>b</sup>Centro Studi Luca D'Agliano, Milan <sup>c</sup>CIFREM, Trento

# Abstract

Differences in productivity across regions have been mainly attributed to agglomeration economies, technology and human capital, while almost no evidence has been provided on the role of internationalization. In this paper we build unique measures of outward and inward foreign direct investment (FDI) counts at the NUTS 2 level and we assess the relationship between regional productivity and foreign investments in Europe. Regions with larger outflows of foreign investments show higher productivity growth, but this correlation fades down with the number of investments and eventually becomes negative in regions with very high outward orientation. Inward investments are also positively associated with regional productivity growth, but only above a certain threshold. Results are robust to the introduction of a number of regional characteristics, to the control for endogeneity of foreign investments, and for spatial dependence.

JEL classification: C23, C26, F23, O47, O52, R11

*Keywords:* Regional productivity, foreign investments, Europe, spatial econometric models, instrumental variables,

<sup>&</sup>lt;sup>\*</sup>This is a subtantially revised version of a paper previously appeared as "The Effect of Foreign Investments on European Regional Productivity", *Quaderni del Dipartimento di Economia, Finanza e Statistica*, n.74, 2010. The authors thank Carlo Altomonte, Antonio Alvarez, Giangranco Di Vaio, Florian Mayneris, Bruno Merlevede, Cristiano Perugini, Alessandro Sembenelli, Hylke Vandenbussche, Francesco Venturini and the participants at the conferences and seminars in Rome (November 2009), Oviedo (December 2009), CORE-Louvain (May 2010), Catania (September 2010), Tubingen (November 2010), Madrid (December 2010). Financial support from the European Community's Seventh Framework Programme (Project INGINEUS, Grant Agreement No.225368, www.ingineus.eu) is greatly acknowledged.

Email addresses: davide.castellani@unipg.it (Davide Castellani), fabio.pieri@unitn.it (Fabio Pieri)

# 1. Introduction

Regional competitiveness and social and economic cohesion have been crucial concerns for policy makers —especially in the European Union  $(EU)^1$ — and have attracted a considerable amount of economic research. In particular, empirical works have focused on explaining differences in productivity among EU regions. Agglomeration economies, technology and human capital have been most often considered as the key dimensions to explain such differences<sup>2</sup>. With the notable exception of Gambardella, Mariani, and Torrisi (2008), internationalization is rarely considered as a factor affecting regional productivity. This is probably due to the lack of accurate measures of a region's openness<sup>3</sup>. This lack of evidence is at odds with the increasing relevance of regions in the global economy, and in Europe in particular. With the free movement of goods, capital and labour, it makes less and less sense to think about economic relations within Europe in terms of the standard paradigm of international trade. One should rather take a regional perspective and emphasize relations of sub-national units within the EU and with the rest of the world (Krugman, 1993). In this work, using a novel dataset on international investment projects, we are able to build unique measures of outward and inward foreign direct investment (FDI) at the regional level (NUTS 2)<sup>4</sup> for the countries of the European Union (EU-27). This allows us to assess – for the first time – the extent to which regional productivity is associated with internationalization, and in particular with foreign investments by multinational enterprises (MNEs). This issue is particularly relevant in the European Union (EU), which is a major home and host area for FDIs: both inward and outward FDIs account for almost 4% of the EU GDP, but with very differentiated patterns across countries<sup>5</sup>. Empirical works have also documented that inward FDIs are not uniformly distributed across regions within individual countries (Head and Mayer, 2004; Basile, Castellani, and Zanfei, 2008). Instead, evidence is lacking on the different propensity of European regions to engage in outward FDIs.

In order to investigate whether foreign investments actually affect regional productivity, we estimate regressions of (one-year) productivity growth as a function of one-year-lagged foreign investments. We find that inward FDIs have a positive and significant effect on regional productivity growth, but this effect is sizable only for a relatively large number of investment projects. Conversely, regions with at least one outward FDI project have higher productivity growth, but the effect of FDIs fades down with the number of projects, and may eventually become negative in regions with very large outward flows. These results are robust to a number of controls. In particular, we have added several regional characteristics (both in level and in first-differences), accounted for spatial dependence and controlled for the possible endogeneity of FDIs.

This piece of evidence bears implications for policy. In particular, it suggests, on the one hand, that fears of hollowing-out as a consequence of outward investments are not entirely founded, and local economies may in fact benefit from the fact that incumbent firms move some production abroad, and, on the other hand, that substantial investments may be needed to generate sizable effects on host country regional productivity growth.

The rest of the paper is organized as follows: Section 2 presents the related literature on the links between foreign direct investments and productivity; Section 3 describes our empirical strategy; Section 4 provides details on the characteristics of the data and focuses on how the main variables of interest have been measured and constructed; Section 5 provides some descriptive evidence, while Section 6 illustrates the econometric results and the robustness checks. Section 7 concludes the paper.

 $<sup>^{1}</sup>$ As a matter of fact, 35% of the EU budget for the period 2007-2013 has been allocated to promote social and economic cohesion among the regions of its member states.

 $<sup>^{2}</sup>$ See, for example, the empirical evidence on EU regions in Ciccone (2002), Paci and Usai (2000).

 $<sup>^{3}</sup>$ In fact, Gambardella, Mariani, and Torrisi (2008) introduce a generic measure of openness using the share of hotels in the population and the share of the population which speaks a second language.

 $<sup>^{4}</sup>$ NUTS is an acronym for Nomenclature of Units for Territorial Statistics which indicates a hierarchical classification of administrative areas used by the European statistical office (Eurostat). NUTS levels (1-3) indicate different degrees of aggregation.

<sup>&</sup>lt;sup>5</sup>For example outward FDIs, as a share of GDP, go from values close to zero in most New Member States, to around 1% in countries such as Italy and Greece and more than 5% in the UK, France and Spain; on the other hand, inward FDIs range from around 1% of GDP in Greece, Italy and Germany, to more than 10% of GDP in Bulgaria, Belgium and Estonia.

# 2. Foreign investments and productivity: theory and evidence

#### 2.1. Theory

From a theoretical point of view, the links between foreign investments and productivity of home and host countries have been investigated extensively, but with inconclusive results. Substantial work have been done regarding the direct and indirect effects of inward FDIs on host economies. Direct effects refer to the fact that incoming multinationals tend be relatively more productive than domestic firms and to concentrate in higher productivity sectors (Barba Navaretti and Venables, 2006). Thus, entry of foreign multinationals changes the composition of the host economy –both within and between sectors– contributing to increase the aggregate productivity. Foreign multinationals may also have indirect effects, inducing pecuniary and technological externalities but also determining a business stealing effect (Görg and Strobl, 2005; Castellani and Zanfei, 2006). While the former usually provide a positive contribution to aggregate productivity, the latter may have opposite effects. To the extent that local firms are less productive than the foreign ones, the business stealing effect, forcing local firms to shrink or exit, may be beneficial for the aggregate productivity. However, if foreign multinationals transfer only the relatively lower value added activities in the host region (such as in the case of offshoring of distribution activities), while domestic firms carried out most of the production process in the region, the crowding-out effect may be detrimental for aggregate productivity dynamics, since exiting firms would account for a larger share of regional value added.

Outward investments have direct and indirect effects on the productivity of the home economy too. As for the direct effects, firms engaging in foreign activities (either through export or foreign investments) are more productive than purely domestic ones, since they need to overcome the cost of doing business abroad. By going abroad, firms can reach larger markets, thus they grow larger and this contributes to increasing aggregate productivity (Helpman, Melitz, and Yeaple, 2004). At the same time, this allows firms to reap the benefit of higher economies of scale and provides further incentives to invest in R&D (Petit and Sanna-Randaccio, 2000). Furthermore, foreign investors may be able to source foreign knowledge (Cantwell, 1995; Fosfuri and Motta, 1999), which will increase their productivity, boost their growth, and contribute to raising aggregate productivity. Admittedly, outward investments may also be associated with a decrease in the size and productivity of home activities. This would occur when domestic firms relocate a substantial share of their activities abroad. In this case, the competitiveness boost may not be able to compensate the loss in terms of value-added resulting from offshoring.

Outward investments may also contribute to increase the aggregate productivity through indirect effects on the performance of local firms. On the one hand, an increase in size, productivity and/or knowledge of home multinationals may spill-over on other domestic firms through input-output relations and imitation. On the other hand, to the extent that investing firms move value-added creating activities, domestic suppliers along the value chain may be forced to shrink or to exit. At the same time, opportunities may arise in upstream or downstream sectors, for example in activities like logistics, R&D, design, and other business services. The overall effect of this process on aggregate productivity may be positive or negative, according to the balance between the productivity of firm entering (or increasing the market share) and exiting the market (or shrinking).

Various theoretical arguments can be used to support that the effects of (inward and outward) FDIs are relatively confined in space and, thus, the regional level would more appropriate than the contry level to capture them. First, the smaller the units of observation, the easier would be to appreciate the direct effects of inward and outward FDI, which may be more diluted in more aggregate data. Second, indirect effects may be enhanced by the geographic proximity, which can be important for transmitting knowledge as faceto-face communication (Audretsch and Feldman, 2004). Third, in the presence of transport costs, vertical linkages (which foster pecuniary and knowledge externalities) occur between closely-located suppliers and customers (Venables, 1996). Finally, to the extent that multinationals serve the local markets, crowding out and business stealing effects are spatially confined. Admittedly, since firms competing with multinationals may not be local companies, these effects are likely to span across regional borders.

#### 2.2. Evidence

Since theoretical results do not predict clearcut effects, the issue of whether foreign direct investments have positive or negative effects on aggregate productivity becomes mainly an empirical question. Applied works on inward FDIs and productivity have provided sound evidence that the entry of MNEs is associated with a positive direct contribution to the productivity of host economies; moreover, multinational firms contribute to changes in the industrial mix towards relatively more knowledge and technology intensive sectors. Evidence on indirect effects is more mixed, and it seems to depend both on the characteristics of the multinational investments and those of firms in the host economy. Econometric evidence on inward FDIs and productivity have been provided mainly with firm-level studies on one (or more) countries and with more aggregate cross-country studies (Barba Navaretti and Venables, 2006). A few empirical works have also taken a regional perspective within individual countries. For example Mullen and Williams (2007) analyze the regional spillovers of FDI in US states, while in Europe, among others, Driffield (2004) and Girma and Wakelin (2007) focus on UK regions, Crespo, Fontoura, and Proença (2009) on Portugal, Altomonte and Colantone (2009) on Romania, Halpern and Murakozy (2007) on Hungary. Most of these studies specifically address the spatial aspects of spillover and find that, indeed, the activity of foreign multinationals affects relatively more the productivity of domestic firms located nearby. Rather surprisingly, despite the increasing competition among local territories both within and across national boundaries to attract foreign investors (Basile, Castellani, and Zanfei, 2008; Blonigen and Kolpin, 2007; Davies, 2005), cross-country evidence of the effects of inward FDIs at a sub-national level is still lacking.

The literature on outward investments and productivity is more scattered, but has gained momentum in the last decade. Many studies in this field have provided evidence that firms investing abroad tend to be more productive than their home country counterparts (Greenaway and Kneller, 2007): these results would predict that in regions with a larger share of highly productive firms (thus a higher average productivity) one would observe a higher number of firms investing abroad. Other studies have found that investing abroad may further reinforce productivity of investing firms (Barba Navaretti, Castellani, and Disdier, 2010; Branstetter, 2006; Debaere, Lee, and Lee, 2010; Griffith, Harrison, and Reenen, 2006), while only a few works in this literature have addressed the indirect effects from firms investing abroad (Castellani and Zanfei, 2006; Vahter and Masso, 2007), finding that the growth of domestic multinationals in the home country can be a source of spillovers for the local firms.

At the aggregate level, few studies have been conducted on the relation between outward FDIs and productivity, and they also show mixed results. For example, van Pottelsberghe de la Potterie and Lichtenberg (2001), in a panel of 13 developed countries, find that outward investments are a more effective channels for international technology transfer among countries with respect to inward FDIs, Driffield, Love, and Taylor (2009) find that outward FDIs are positively related to productivity growth in UK, while Bitzer and Görg (2009), who examine the effect of outward and inward FDIs on domestic total factor productivity for 17 OECD countries, report that only the latter are positively related to a country productivity. To the best of our knowledge there are no studies at the sub-national level regarding the effects of outward FDIs on the productivity of local economies.

#### 3. The empirical model

In order to assess the effect of inward and outward foreign direct investments on regional productivity we specify the following econometric model:

$$y_{ij,t} = \gamma_{OUT} OFDI_{ij,t-1}^{stock} + \gamma_{INW} IFDI_{ij,t-1}^{stock} + \beta k l_{ij,t} + \mathbf{x}_{ij,t} \delta + \mu_i + t \cdot \eta_j + \tau_t + \epsilon_{ij,t}, \tag{1}$$

where  $y_{ij,t}$  is the (log of the) labour productivity of the *i*th region in the *j*th country at time *t*, and  $OFDI_{ij,t-1}^{stock}$  and  $IFDI_{ij,t-1}^{stock}$  are, respectively, (log of) the stocks of outward and inward foreign direct investments in the *i*th region at the t-1 time period. We make the hypothesis that foreign direct investments affect productivity with one-year lag<sup>6</sup>. We include a set of regional characteristics that economic theory has

 $<sup>^{6}</sup>$ This is explicitly tested against the hypothesis that FDI have a contemporaneous effect on productivity in Section 6.

indicated as determinants of productivity and which are likely to be correlated with inflows and outflows FDI in European regions. Thus, the model is augmented with  $kl_{ij,t}$ , which indicates the (log of the) capitallabour ratio and  $\mathbf{x}_{ij,t}$ , which is a vector of (the log of) other regional characteristics, such as the level of human capital, the stock of technological capital, the regional industrial composition and the degree of concentration/diversification of the regional industry<sup>7</sup>. We include a vector of regional effects,  $\mu_i$ , to control for unobserved (and time invariant) regional characteristics which could be correlated both with the stocks of foreign direct investments (incoming or outgoing from the region) and with the regional productivity; a vector of time effects,  $\tau_t$ , to control for factors affecting all regions in the same way in a given year; while the interaction  $t \cdot \eta_j$  is introduced in order to capture the country-specific trends in labour productivity, but also the growth rate (Nicoletti and Scarpetta, 2003). First-differencing equation 1 wipes out the regional fixed effect and leaves us with differences in FDI stocks on the right hand side. This is a handy solution in our case since, as we will discuss in the next section, we do not have information on FDI stocks, because of a constraint on the available data.

The first differenced equation can be written as

 $\Lambda$ 

$$\Delta y_{ij,t} = \gamma_{OUT} \Delta OFDI_{ij,t-1}^{stock} + \gamma_{INW} \Delta IFDI_{ij,t-1}^{stock} + \beta \Delta k l_{ij,t} + \Delta \mathbf{x_{ij,t}} \delta + \eta_j + \tau_t + \Delta \epsilon_{ij,t}, \tag{2}$$

where  $\Delta$  indicates the difference between the variable at time t and the variable at time t-1. With respects to the variables measuring foreign direct investments, differences are computed between the variable at time t-1 and the variable at time t-2.

The relationship between investments stocks and flows can be formalized, with some approximation<sup>8</sup> as

$$\Delta_{(t-1,t-2)}OFDI_{ij}^{stock} \cong OFDI_{t-1}^{flows},\tag{3}$$

and

$$A_{(t-1,t-2)}IFDI_{ij}^{stock} \cong IFDI_{t-1}^{flows}.$$
(4)

Plugging 3 and 4 into 2 yields

$$\Delta y_{ij,t} = \gamma_{OUT} OFDI_{ij,t-1}^{flows} + \gamma_{INW} IFDI_{ij,t-1}^{flows} + \beta \Delta k l_{ij,t} + \Delta \mathbf{x_{ij,t}} \delta + \eta_j + \tau_t + \Delta \epsilon_{ij,t}.$$
(5)

Equation 5 has an appealing interpretation in our case: the parameters  $\gamma_{OUT}$  and  $\gamma_{INW}$  explicitly consider the relationship between outward and inward flows of investments and the growth rate of the labour productivity.

# 4. Data and variables

#### 4.1. Data sources

We exploit an original database, which has been compiled recovering data from different sources. Data refer to European regions, at the NUTS 2 level: this level of analysis has been chosen for three main reasons. First of all, it is suitable for taking into account the within-country heterogeneity (in terms of labour productivity, foreign direct investments and the other observed and unobserved characteristics); second, it allows for comparable units across different countries; finally, more information is available on other regional characteristics at this level of disaggregation.

Information on regional gross value added come from the EU Regional Database developed and maintained by Eurostat<sup>9</sup>, while data on employment and capital investments at the regional level come from

 $<sup>^{7}</sup>$ The choice of the control variables is based on previous theoretical and empirical works. We cross-refer the reader to the Appendix A for a detailed discussion on the control variables and their measurement.

 $<sup>^{8}</sup>$ The approximation is due to the fact that change in the stock is given by the flow of investments plus the depreciation of the existing capital stock. Unfortunately the lack of the stock of investments forces us to rely on the approximation illustrated in the text.

<sup>&</sup>lt;sup>9</sup>See the Eurostat web page

http://epp.eurostat.ec.europa.eu/portal/page/portal/region\_cities/.

the European Regional Database, developed by Cambridge Econometrics (release 2006). We have used these information in order to build a measure of labour productivity and a measure of the capital-labour ratio at the regional level. Data on outward and inward FDIs, come from *fDi Markets* an online database maintained by fDi Intelligence — a specialist division of the Financial Times Ltd—, which monitors crossborder investments covering all sectors and countries worldwide. Relying on media sources and company data, fDi Markets collects detailed information on cross-border greenfield investments (available since 2003). fDi Markets data are based on the announcement of the investment and provides daily updated data. For each FDI project, fDi Markets reports information on the investment (e.g., the leading industry sector of the investment), the home and host countries, and regions and cities involved, and the investing company (e.g., location, parent company). The database is used as the data source for FDI project information in UNCTAD's World Investment Report and in publications by the Economist Intelligence Unit.

#### 4.2. Labour Productivity

The dependent variable is the labour productivity, which has been computed as the ratio of the regional gross valued added (at basic prices in millions of euro) obtained from the Regio database, to employment (thousands) in in each region, which has been recovered from the European Regional Database. Given that the price indexes for the gross value added are not available at the regional level, the nationwide indexes, which are available in the *Growth and Productivity Accounts* database developed by EU KLEMS<sup>10</sup> (releases 2008 and 2009), were used to deflate the value added. The last year for which information on value added are available in the Regio database is 2006. The variable has been included in logs in the performed econometric analysis,  $y_{ijt}$ .

#### 4.3. Foreign investments

Data on inward and outward foreign direct investments flows  $(IFDI_{ijt}^{flows}, OFDI_{ijt}^{flows})$  have been recovered from the fDi Markets database. This source tracked 60,301 worldwide investments projects appeared on publicity available information sources in the period 2003-2008<sup>11</sup>. One of the limitations of the fDi Market database is that it collects planned future investments. Some of these projects may not actually be realized or may be realized in a different form from the one originally announced. However, the database is regularly updated and projects which have not been completed are deleted from the database. In this regards, data on the projects related to the early years of the series should be more reliable than data regarding the last years of the series. We tackle this issue by dropping the last two years of data, so we use information on FDI from 2003 to 2006. Our measures of FDI flows is then built as the number of inward/outward investment projects in/from each region in each year of the period 2003-2006:

$$wFDI_{ijt}^{flows} = \#$$
of projects in region *i* belonging to country *j*, in year *t*,

where  $w = \{I, O\}$ , are respectively inward and outward investments.

Admittedly, the count of FDI projects may not be an accurate proxy of FDI flows, since it does not weights investments for the value of the capital involved. However, the correlation coefficients (0.82 and 0.83), reported in Table 1, between the distribution of FDI projects by EU countries and the actual distribution of FDI flows, as reported by UNCTAD, reassures us that data on investment projects are actually a good proxy for FDI flows. As expected, almost 90% of EU outward investments are made from EU-15 countries, while inward investments are split more evenly among EU-15 and EU-12 countries: United Kingdom, Germany and France result to be the leading countries both in terms of inward and outward FDIs in the period which goes from 2003 to 2006. As for the inward investments, Poland, Romania, Hungary, Czech Republic and Bulgaria show a good performance.<sup>12</sup>.

 $<sup>^{10}{\</sup>rm See}$  the web page of the EU KLEMS project at http://www.euklems.net/

<sup>&</sup>lt;sup>11</sup>A team of in-house analysts search daily for investment projects from various publicly available information sources, including, Financial Times newswires, nearly 9,000 media, over 1,000 industry organisations and investment agencies, data purchased from market research and publication companies. Each project identified is cross-referenced against multiple sources,

	Outw	ard		Inwa	rd
Country	∦ proj.	flows	Country	∦ proj.	flows
Germany	22.2	11.7	United Kingdom	16.0	25.8
United Kingdom	20.3	16.3	France	9.2	15.2
France	13.8	17.6	Germany	8.3	8.1
Italy	6.3	5.7	Poland	6.5	3.0
Netherlands	5.9	13.7	Spain	6.2	7.2
Sweden	5.9	4.7	Romania	5.9	1.7
Austria	5.1	2.0	Hungary	5.4	1.4
Spain	4.6	11.7	Czech Republic	4.1	1.5
Finland	3.1	0.3	Bulgaria	4.1	1.1
Belgium	2.5	7.9	Ireland	4.1	-1.6
Denmark	1.9	1.4	Italy	3.9	5.9
Ireland	1.4	2.7	Sweden	3.2	3.4
Slovenia	1.1	0.1	Netherlands	3.1	5.1
Greece	0.9	0.4	Belgium	2.9	10.8
Latvia	0.9	0.0	Slovakia	2.6	0.8
Estonia	0.6	0.1	Lithuania	2.4	0.2
Portugal	0.5	1.2	Austria	2.2	1.9
Luxembourg	0.5	1.0	Denmark	1.9	1.2
Poland	0.5	0.7	Latvia	1.7	0.2
Czech Republic	0.5	0.1	Estonia	1.5	0.4
Hungary	0.4	0.4	Portugal	1.3	1.5
Lithuania	0.4	0.0	Greece	1.1	0.6
Cyprus	0.2	0.1	Finland	0.9	1.2
Romania	0.2	0.0	Slovenia	0.8	0.2
Slovakia	0.1	0.0	Luxembourg	0.4	2.7
Bulgaria	0.1	0.0	Cyprus	0.3	0.3
Malta	0	0.0	Malta	0.2	0.2
Total	100	100		100	100
Pearson corr. coefficient	0.8	2		0.8	3

Table 1: fDi Markets projects vs. UNCTAD Flows, 2003-2006

Unfortunately, official statistics on inward and outward investments at the regional level are not available, so we cannot benchmark fDi Markets data as this finer geographical level. However, we can check the data against previous results and some theoretical expectations. To this end, we will exploit the visual representation of the geographical distribution of the number of investment projects at the NUTS 2 level, provided in Figure 1. In line with previous evidence on the role of agglomeration economies for the location of multinational firms (e.g. Crozet, Mayer, and Mucchielli (2004); Bobonis and Shatz (2007)), inward and outward investments appear highly concentrated in a limited number of clustered regions within each country, including the regions around the major cities. In the subsequent econometric analysis, we will assess to what extent this within-country heterogeneity in inward and outward investments maps into different productivity dynamics. A closer inspection of the maps in Figure 1 reveals that outward investments are concentrated in some of the core regions of Continental Europe and the UK, while inward investments are also frequent in a number of peripheral areas, such as the Eastern European countries, Ireland, Scotland and Andalusia in Spain. The latter result is consistent with previous evidence on the positive role of EU Structural and Cohesion Policies in attracting FDI in peripheral regions (Basile, Castellani, and Zanfei, 2008).

# 5. Descriptive analysis

The time structure of our data imposes some constraints on the empirical analysis. In particular, regional productivity is observed only up to 2006, while information on foreign investments are available for the period 2003-2008. Thus, if we want to assess the econometric relationship between the latter and the former, we are left with four years of data: 2003, 2004, 2005 and 2006. Due to the lack of the information regarding some regional characteristics, regions belonging to Norway, Switzerland and Denmark cannot be considered<sup>13</sup>.

Table 2 provides some basic statistics for the variables used in the econometric analysis. As concerns foreign investments, Table 2 shows that, on average, from each region about 14 outgoing investments and 10 incoming investments per year have been recorded. However, the distribution of the number of investments is highly skewed: from more than 25% of regions no outward investment in one year would originate and more than 10% would not attract any inward investment.

The skewness of the foreign investments variables induces us to model their effect as a combination of a dummy taking value equal to '0' for those observations (region/year) where no investments have taken place and a continuous variables taking the value equal to the log of the number of investments in the case of non-zero investments, and '0' otherwise<sup>14</sup>. In other words, investments variables enter the regressions as follow:

$$wFDI(d)_{i,t} = \begin{cases} = 1 & \text{if } \# \text{ of investments } \underset{i,t}{w} > 0 \\ = 0 & \text{if } \# \text{ of investments } \underset{i,t}{w} = 0 \end{cases}$$

$$wFDI(log)_{i,t} = \begin{cases} = log(\# \text{ of investments } \underset{i,t}{w}) & \text{if } \# \text{ of investments } \underset{i,t}{w} > 0 \\ = 0 & \text{if } \# \text{ of investments } \underset{i,t}{w} = 0 \end{cases}$$

where  $w = \{I, O\}$  are respectively inward and outward investments. This specification allows to distinguish the effect of a region being generally involved in the internationalization process, which is captured by the dummy variable, from the effect of the degree of internationalization, which is captured by the continuous variable in logs.

and over 90% of projects are validated with company sources. More information at http://fdimarkets.com/

 $<sup>^{12}</sup>$ A careful inspection reveals that the number of projects overestimates inward FDIs to some New Member States, such as Poland, Romania, Bulgaria, Hungary and Czech Republic, probably due to the fact that these countries received a large number of project of relatively small-scale investments project

 $<sup>^{13}</sup>$ See Table 13 in the Appendix for the detailed list of regions, that have been considered in the econometric analysis.

 $<sup>^{14}</sup>$ We take the log of the number of investments so that we can interpret the coefficient of the continuous variable as an elasticity.



Figure 1: Regional distribution of international investment projects, 2003-2006

(a) Inward investments



(b) Outward investments

Variable	Notation	Unit	Count	Obs.	Mean	Std. Dev	p10	p25	p50	p75	p90
Outward FDI	OFDI	count	14135	1032	13.697	39.710	0	0	3	11	30
Inward FDI	IFDI	count	10802	1032	10.467	19.825	0	1	4	12	24
Labour productivity	y	ratio (log)		1017	3.360	0.751	1.956	3.202	3.651	3.856	3.948
Capital-labour ratio	kl	ratio (log)		1036	4.148	0.863	2.714	3.949	4.387	4.753	4.923
Human capital	h cap	ratio (log)		1010	-1.468	0.378	-2.040	-1.728	-1.403	-1.189	-1.035
Herfindahl index	hhi	formula (log)		922	-1.377	0.177	-1.602	-1.514	-1.391	-1.246	-1.144
Innovation stock	tech	formula (log)		1036	-0.992	1.859	-3.721	-2.360	-0.416	0.397	0.982
Share of other industies	$SH_{EF}$	share		922	0.089	0.023	0.062	0.072	0.084	0.101	0.119
Share of High-tech man.	SH_HT	share		922	0.066	0.035	0.028	0.043	0.060	0.084	0.112
Share of Low-tech man.	SH_LT	share		922	0.125	0.046	0.068	0.088	0.122	0.153	0.191
Share of KI svcs	SH_KIS	share		922	0.316	0.088	0.212	0.254	0.309	0.379	0.431
Share of LKI svcs	SH_LKIS	share		922	0.336	0.047	0.280	0.312	0.338	0.364	0.392
Labour productivity-growth rate	$\Delta_{(t,t-1)}y$	ratio (log, differences)		1017	0.020	0.044	-0.017	0.003	0.017	0.034	0.059
Capital-labour ratio-growth rate	$\Delta_{(t,t-1)}k$	ratio (log, differences)		1036	0.022	0.026	-0.003	0.007	0.018	0.032	0.053
Human capital-growth rate	$\Delta_{(t,t-1)}hcap$	ratio (log, differences)		1002	0.039	0.072	-0.037	-0.002	0.036	0.074	0.119
Herfindahl index-growth rate	$\Delta_{(t,t-1)} hhi$	formula (log, differences)		891	0.009	0.035	-0.033	-0.009	0.008	0.028	0.048
Innovation stock-growth rate	$\Delta_{(t,t-1)} tech$	formula (log, differences)		1036	0.047	0.156	-0.079	-0.018	0.033	0.081	0.174
Share of other industries-growth rate	$\Delta_{(t,t-1)}$ SH_EF	share (differences)		891	0	0.009	-0.010	-0.004	0.001	0.006	0.012
Share of High-tech mangrowth rate	$\Delta_{(t,t-1)}$ SH_HT	share (differences)		891	0	0.009	-0.011	-0.006	-0.001	0.004	0.009
Share of Low-tech mangrowth rate	$\Delta_{(t,t-1)}$ SH_LT	share (differences)		891	0	0.011	-0.016	-0.009	-0.002	0.003	0.009
Share of KI svcsgrowth rate	$\Delta_{(t,t-1)}$ SH_KI	share (differences)		891	0	0.015	-0.012	-0.003	0.004	0.013	0.022
Share of LKI svcsgrowth rate	$\Delta_{(t,t-1)}$ SH-LKIS	share (differences)		891	0	0.016	-0.017	-0.009	0.002	0.010	0.020

Table 2: Descriptive statistics, 2003-2006

Figure 2 provides a graphical representations of the variables measuring the labour productivity in levels and growth at the NUTS 2 level. Labour productivity, are clearly higher in the core regions of the EU-15, while decline in Southern European regions and reach minimum values in the regions of EU-12 countries. As for the growth rates, rather similar patterns are observed in regions belonging to the same country mainly in EU-12 countries, but also in Italy, France and Spain; while in Germany and UK productivity growth displays a remarkable within-country variability. These insights are confirmed in Tables 10 and 11 reported in the Appendix, which present descriptive statistics by country. In order to account for possible biases stemming from these country patterns in productivity growth, country dummies will introduced in our estimated equation, as illustrated in Equation 5.

#### 6. Econometric analysis

#### 6.1. Baseline results

Following the specification of FDI variables in Section 5, the estimated model becomes:

$$\Delta y_{ij,t} = \alpha + \beta \Delta k l_{ij,t} + \Delta \mathbf{x}_{ij,t} \delta + + \gamma_O^d OFDI(d)_{ij,t-1} + \gamma_O^{log} OFDI(d)_{ij,t-1} \cdot OFDI(log)_{ij,t-1} + + \gamma_I^d IFDI(d)_{ij,t-1} + \gamma_I^{log} IFDI(d)_{ij,t-1} \cdot IFDI(log)_{ij,t-1} + + \eta_j + \tau_t + \Delta \epsilon_{ij,t}.$$
(6)

We estimate Equation 6 by OLS, and the results are reported in Table 3. In this case we are left with three pooled cross-sections of first-differenced equations: 2004-2003, 2005-2004 and 2006-2005<sup>15</sup>. In this and the following regressions we report robust standard errors clustered by regions to control for the lack of independence of observations referring to the same region over time<sup>16</sup>.

In Specification (1), we look at the effects of inward and outward foreign direct investments (made in year t-1) on productivity growth rates, taking into account the change in the capital-labour ratio but without controlling for the other regional characteristics (i.e. human capital, technological capital, the industrial mix and its degree of concentration/diversification). Coefficient of the variables related to inward FDIs,  $\gamma_I^d$ and  $\gamma_I^{log}$ , suggest that for low levels of incoming investments the effect on regional productivity is negative, because the value of the coefficient of the dummy variable dominates the coefficient of the continuous variable. However, the effect of outward FDIs increases as the number of incoming projects becomes larger: in other words, inward FDIs have a positive effect on regional productivity, above a threshold number of investments. On the other hand, outward FDIs have a positive effect on regional productivity,  $\gamma_O^d$ , but the effect decreases as the number of outward investments increases, as captured by  $\widehat{\gamma_O^{log}}$ . In Specification (2) the change in the quality of the industrial mix is taken into account, together with changes in the level of human capital, in the technological capital stock and in the degree of concentration/diversification of the industrial mix. A non-negligible loss in the sample size occurs from Specification (1) to (2), and this is mainly due to the lack of data for sectoral employment shares in several regions: these missing values bring to corresponding loss of usable observations in the industrial mix variables  $(SH_{s*ijt})$  and in the Herfindahl-Hirschman index  $(HHI_{ijt})^{17}$ . To a lesser extent, few missing values are in the variables measuring the level of human capital and the technological capital. Despite the sizable reduction in sample size, results

 $<sup>^{15}</sup>$ It is worth mentioning that it would be highly desirable to specify differences longer than one-year for productivity growth but, given the short time span available in our data, this would reduce the number of observations, thus increasing measurement errors and reducing the precision of our estimates.

 $<sup>^{16}</sup>$ All the regressions have been estimated using Stata 10.1, except for those in Section 6.2.2, which have been run using the environment R.

<sup>&</sup>lt;sup>17</sup>Data for employment shares are not available for the following regions in some (or all) of the three waves of growth rates: Belgium (BE34), Germany (DE30, DE41, DE42, DE50, DE60, DEB2, DED3, DEE0) Denmark (all regions; DK01, DK02,



Figure 2: Regional patterns of labour-productivity level and growth, 2003-2006 (average)

(a) Labour productivity (level)



(b) Labour productivity (growth)

$\begin{array}{c c c c c c c c c c c c c c c c c c c $				Specification	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $			1	2	3
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Variable	Coefficient			
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$OFDI_{t-1}$ (dummy)	$\gamma_{O}^{d}$	0.0088***	$0.0076^{***}$	0.0075**
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		.0	(0.0029)	(0.0029)	(0.0029)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$OFDI_{t-1}(log. of n.inv)$	$\gamma_{O}^{log}$	-0.0030***	-0.0027***	-0.0029***
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	,	,0	(0.0009)	(0.0009)	(0.0010)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$IFDI_{t-1}(dummy)$	$\gamma_I^d$	-0.0074***	-0.0024	-0.0072***
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		1	(0.0026)	(0.0025)	(0.0027)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$IFDI_{t-1}(\log of n.inv)$	$\gamma_I^{log}$	0.0031***	0.0020*	0.0031***
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		1	(0.0011)	(0.0011)	(0.0012)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\Delta_{t,t-1}kl$	β	0.2401***	0.3592***	0.2392***
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	0,0 1	,	(0.0839)	(0.1088)	(0.0842)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\Delta_{t,t-1}$ hcap	$\delta_{hcap}$	· · · ·	-0.0120	0.0003
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	-,			(0.0164)	(0.0137)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\Delta_{t,t-1}$ hhi	$\delta_{hhi}$		$0.1975^{***}$	0.1577**
$\begin{array}{cccc} \Delta_{t,t-1} \text{tech} & \delta_{tech} & -0.0001 & 0.0008 \\ & (0.0083) & (0.0100) \\ \Delta_{t,t-1} \text{SH}\_\text{FF} & \delta_{EF} & 0.0420 & 0.1434 \\ & (0.1434) & (0.1509) \\ \Delta_{t,t-1} \text{SH}\_\text{HD} & \delta_{HD} & 0.0910 & 0.1638 \\ & (0.1381) & (0.1416) \\ \Delta_{t,t-1} \text{SH}\_\text{LD} & \delta_{LD} & -0.1648 & -0.1430 \\ & (0.1438) & (0.1557) \\ \Delta_{t,t-1} \text{SH}\_\text{KIS} & \delta_{KLG} & = 0.3420^{**} & = 0.1876 \end{array}$	- ,			(0.0616)	(0.0740)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\Delta_{t,t-1}$ tech	$\delta_{tech}$		-0.0001	0.0008
$\begin{array}{cccccccccccccccccccccccccccccccccccc$				(0.0083)	(0.0100)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\Delta_{t,t-1}$ SH_EF	$\delta_{EF}$		0.0420	0.1434
$\begin{array}{cccc} \Delta_{t,t-1}\text{SH}\_\text{HD} & \delta_{HD} & 0.0910 & 0.1638 \\ & & & & & & \\ (0.1381) & (0.1416) \\ \Delta_{t,t-1}\text{SH}\_\text{LD} & \delta_{LD} & -0.1648 & -0.1430 \\ & & & & & & \\ & & & & & & \\ (0.1438) & (0.1557) \\ \Delta_{t,t-1}\text{SH}\_\text{KIS} & \delta_{KLG} & = 0.3420^{**} & = 0.1876 \\ \end{array}$				(0.1434)	(0.1509)
$\begin{array}{cccc} & (0.1381) & (0.1416) \\ \Delta_{t,t-1}\text{SH}\text{LD} & \delta_{LD} & & -0.1648 & -0.1430 \\ & & (0.1438) & (0.1557) \\ \Delta_{t,t-1}\text{SH}\text{KIS} & \delta_{KLG} & & -0.1876 \end{array}$	$\Delta_{t,t-1}$ SH_HD	$\delta_{HD}$		0.0910	0.1638
$\begin{array}{cccc} \Delta_{t,t-1}\text{SH}\text{LD} & \delta_{LD} & -0.1648 & -0.1430 \\ & & & & & & \\ & & & & & & \\ & & & & $				(0.1381)	(0.1416)
$\begin{array}{c} (0.1438) & (0.1557) \\ \Delta_{444} \text{ SH KIS} & \delta_{556} & -0.3420^{**} & -0.1876 \\ \end{array}$	$\Delta_{t,t-1}$ SH_LD	$\delta_{LD}$		-0.1648	-0.1430
$\Lambda_{VV}$ SH KIS $\delta_{VVG}$ =0.3420** =0.1876				(0.1438)	(0.1557)
$\Delta_{t,t-10}$	$\Delta_{t,t-1}$ SH_KIS	$\delta_{KIS}$		$-0.3420^{**}$	-0.1876
(0.1325) $(0.1690)$				(0.1325)	(0.1690)
$\Delta_{t,t-1}$ SH_LKIS $\delta_{LKIS}$ -0.4560*** -0.3052*	$\Delta_{t,t-1}$ SH_LKIS	$\delta_{LKIS}$		-0.4560***	-0.3052*
(0.1417) $(0.1751)$				(0.1417)	(0.1751)
Constant $\alpha$ 0.0272*** 0.0212*** 0.0270***	Constant	$\alpha$	$0.0272^{***}$	$0.0212^{***}$	$0.0270^{***}$
(0.0039) $(0.0039)$ $(0.0039)$			(0.0039)	(0.0039)	(0.0039)
Country dummies $\eta_j$ Yes Yes Yes	Country dummies	$\eta_j$	Yes	Yes	Yes
Year dummies $ au_t$ Yes Yes Yes	Year dummies	$ au_t$	Yes	Yes	Yes
Observations 755 659 746	Observations		755	659	746
Regions 258 237 255	Regions		258	237	255

Table 3: Econometric results - Baseline (OLS)

Significance levels: \* 10%, \*\* 5%,\*\*\* 1% Cluster-robust standard errors in brackets

on coefficients of outward foreign investments do not change much, while in the case of incoming projects, the coefficient on dummy variable,  $\widehat{\gamma_I^d}$ , slightly drops becoming non significantly different from zero and the coefficient of the number of projects,  $\widehat{\gamma_I^{log}}$ , becomes poorly significant, even if the coefficient is rather stable in magnitude. The observed changes in the coefficients are the result of the sample-selection due to missing values in sectoral employment shares. This fact is confirmed by Specification (3), in which we have filled in most of the missing values in the vector  $\mathbf{x_{ii,t}}^{18}$ .

Results from Specification (3) are in line with those of Specification (1). The result on inward investments is a slightly sensitive to choice of the sample under analysis rather than to the introduction of further controls. Overall, Specification (3) is our favorite one, because it allows to control for an important set of regional characteristics without reducing the sample size. The cost for this choice is the use of variables with some imputed values for a limited number of observations: that is, the effect of the regional characteristics which we use as controls could not be always consistent. However, there is no reason to think that this should affect the sign and the magnitude of the coefficients related to inward and outward FDIs variables. By the way, most of the coefficients of the controls result to be not significant for explaining the regional productivity growth. In particular, neither the contemporaneous change in the human capital, nor the change in the technology capital –even if they show the expected signs– seem to significantly explain the regional differences in productivity growth. However, the vector of controls is jointly significant, as reported in the first row of Table 4. In Table 4 (second row) we report a test for the joint significance of foreign direct

Table 4: Tests on parameters of the baseline Specification (3)

Null Hypothesis (H0)	Conditions	F-Statistics	Critical value (5%)
No regional characteristics effects	$\beta = \delta' = 0$	2.92	1.92
No FDIs effects	$\gamma_w^{log} = \gamma_w^d = 0$	3.52	2.41
No country dummies effects	$\eta' = 0$	119.65	1.56

investments variables: the null hypothesis of no effect by inward and outward foreign direct investments flows is tested and rejected. This confirms the significant role played by foreign direct investments in explaining differences in regional growth rates, once a large set of regional characteristics together with unobserved country-specific trends in productivity have been taken into account. In the third row of Table 4, an F-test on the joint significance of country effects is carried out. The evidence of national trends in labour productivity captured by the national effects is clear: the country dummies result to be jointly significant and failing to account for them would bring us to neglect the significant national patterns of growth, also emerging from Figure 2. It is worth mentioning that after controlling for such country effects, regional differences in the growth in patents and in human capital do not appear to be correlated to the regional productivity growth, while the correlation with international orientation is still significant, although rather small in magnitude<sup>19</sup>.

Finally, let us comment on the threshold effects of inward and outward investments. From Equation 6, the marginal effect of an inward or outward investment on regional productivity growth can be computed

DK03), Spain (ES43), Finland (FI20), France (FR83), Greece (GR11, GR13, GR21, GR22, GR23, G25, GR42, GR43), Italy (ITC2), Netherlands (NL23), Poland (—just for the growth rate 2004-2003— all regions; PL11, PL12, PL21, PL22, PL31, PL32, PL33, PL34, PL41, PL42, PL43, PL51, PL52, PL61, PL62, PL63), Portugal (PT15), United Kingdom (UKE2, UKF3, UKK3, UKK4, UKM5, UKM6).

 $<sup>^{18}</sup>$ We have imputed the missing values in two steps. First, for the period 2002-2006, we assumed that missing values were equal to 'the last or the first available data' in the series. In other words, if an observation was missing in a given region in 2004 but it was observable in 2003, the value in 2004 was set equal to that of 2003. On the other hand, if the observation was missing in a given region in 2002 but it was observable in 2003, the value in 2003, the value in 2003, the value in 2002 was set equal to that of 2003. Thus, we assumed 'zero-changes' were information was not available. Second, in the cases where no data was available or a given region throughout the 2003-2006 period, we imputed using national averages.

<sup>&</sup>lt;sup>19</sup>The derivative of regional productivity with respect to outward investment, evaluated at the median number of projects, is 0.43%, which amounts to one-forth of the median productivity growth (1.7%), when the  $75^{t}h$  percentile is 3.4%. Instead, the effect of inward investments at the median is -0.29%.

as:

$$\frac{\partial \Delta y}{\partial wFDI} = \gamma_w^d + \gamma_w^{log} \cdot log(wFDI).$$
<sup>(7)</sup>

The marginal effect of one more investment will be positive as long as

$$log(wFDI) > \frac{-\gamma_w^d}{\gamma_w^{log}}.$$
(8)

In particular, taking Specification (3) as a reference, with  $\widehat{\gamma_I^d} = -0.0072$  and  $\widehat{\gamma_I^{log}} = 0.0031$ , the marginal effect of receiving one more inward investment would be positive for a number of investments greater or equal than  $\exp \frac{0.0072}{0.0031} = 10.2$ . For outward investments, with  $\widehat{\gamma_O^d} = 0.0075$  and  $\widehat{\gamma_O^{log}} = -0.0029$ , the marginal effect will be positive up to  $\exp \frac{-0.0075}{-0.0029} = 13.3$  investments.





Figure 3 allows to appreciate the extent to which inward and outward investments contribute to productivity growth of EU regions. The Figure plots the cumulative distribution of region/year observations by the number of inward and outward FDIs. The first thing to notice is that outward FDIs are 'twice more rare' than inward FDIs: 28% of region/year observations have zero outgoing projects, as opposed to only 14% in the case of incoming investments. However, there is a sizable number of cases with a rather large number of outward investments, so that the cumulative distributions for OFDI and IFDI cross at 13 investments. Recalling that the threshold level of investments above which the effect is positive is 10.2, from Figure 3 we gather that approximately 30% of region/year observations are above this threshold, and benefit from inward investments. In the case of outward investments, 28% of regions would increase their productivity growth by 0.75% making one project abroad, while about 22% are above the 13.3 threshold, and have thus lower productivity growth then non-internationalized. The remaining 50% are actually experiencing higher productivity growth, thanks to their international orientation.

#### 6.2. Robustness checks

In the previous section we have argued that both inward and outward foreign investments can be a key determinant of differences in productivity growth among the European regions. In the present section we will show that these results are robust to various specifications, and that are not significantly affected by spatial dependence, endogeneity and omitted variables.

6.2.1. Different specifications of the production function and of the effect of contemporaneous investments

Our model is based on a specification of labour productivity regressed on the capital-labour ratio. This is a relatively common specification for empirical analyses on regional data, and is basically equivalent to assuming a Cobb-Douglas production function with constant returns to scale. In this section, we relax the assumption of CRS and estimate the following regression:

$$\Delta v a_{ij,t} = \alpha + \beta_k \Delta k_{ij,t} + \beta_l \Delta l_{ij,t} + \Delta \mathbf{x}_{ij,t} \delta + + \gamma_O^d OFDI(d)_{ij,t-1} + \gamma_O^{log} OFDI(d)_{ij,t-1} \cdot OFDI(log)_{ij,t-1} + + \gamma_I^d IFDI(d)_{ij,t-1} + \gamma_I^{log} IFDI(d)_{ij,t-1} \cdot IFDI(log)_{ij,t-1} + + \eta_j + \tau_t + \Delta \epsilon_{ij,t}.$$
(9)

where  $va_{ij,t}$  is (the log of) gross regional value added, while  $k_{ij,t}$  and  $l_{ij,t}$  denote (the log of) the stock of capital and the total employment in the region. Results, reported in the third column of Table 5, confirm the baseline estimates of Column (3) in Table 3:  $\gamma_O^d$  and  $\gamma_I^d$  are slightly larger in absolute value, while  $\gamma_O^{log}$  and  $\gamma_I^{log}$  are smaller. This changes the thresholds: with those estimates the effect of OFDI would be negative only for regions with more than 35.63 outgoing projects (i.e. less than 10% of the sample), while the effect of IFDI would be positive for regions with more than 15.76 incoming projects (i.e. slightly less than 25% of regions). One should notice that the unconstrained specification of the production function yields unplausibly low returns on capital and labour and significantly decreasing returns to scale. This is probably due to the well known downward bias when estimating production functions in first-differences (Griliches and Mairesse, 1995). This leads us to prefer our baseline estimates.

A further control on the specification concerns our hypothesis that foreign direct investments (both inward and outward) would affect productivity growth with a one-year lag. In order to support this hypothesis, we estimate two additional specifications: the first one with the variables regarding contemporaneous investments only, and the second one with both lagged and contemporaneous investments.

Results, which are reported in Table 5, definitely support our *a priori*: as highlighted in specification  $(3 \lfloor ag1)$ , contemporaneous investments do not have significant correlation with regional productivity growth, except for a small effect of inward investments. Moreover, in the third Column of Table 5, once we introduce both contemporaneous and lagged investments, only the the latter have a significant effect on productivity growth, and the magnitude of the coefficients does not change significantly. It is noting that the specification with lagged investments is also more robust to endogeneity problems: if shocks to current productivity growth would also determine a larger flows of inward and outward investment projects, Specification  $(3 \lfloor ag1)$  may be more sensitive to the simultaneity issue and the use of lagged investments should lessen this problem. We will get back to the issue of endogeneity later in this section.

#### 6.2.2. Accounting for spatial dependence

In our baseline estimation we implicitly assumed that spatial interactions among regions are fully captured by the inclusion of country fixed effects. However, this assumption would hold only if the spatial regional effects were time invariant and specific to each country, i.e. limited to regions belonging to the same country and affecting all the regions belonging to the same country with the same intensity. This is partially confirmed by the maps in Figure 2 and by the high significance of the country dummies in Table 4, but it could be a too restrictive assumption. First, spatial interactions could occur also among regions which belong to different countries; second, they can be time-variant; third, benefits from being localized nearer to

		Specification			
		3	3_pf	3_lag1	3_lag2
Variable	Coefficient				
$OUT(dummy)_{t-1}$	$\gamma^d_O$	$0.0075^{**}$	$0.0085^{***}$		$0.0097^{***}$
		(0.0029)	(0.0029)		(0.0037)
OUT(log. of n.inv) $_{t-1}$	$\gamma_O^{log}$	-0.0029***	$-0.0024^{**}$		-0.0036*
	, i i i i i i i i i i i i i i i i i i i	(0.0009)	(0.0010)		(0.0018)
$INW(dummy)_{t-1}$	$\gamma_I^d$	$-0.0072^{***}$	$-0.0081^{***}$		-0.0067**
		(0.0027)	(0.0026)		(0.0029)
INW(log. of $n.inv$ ) <sub>t-1</sub>	$\gamma_I^{log}$	$0.0031^{***}$	0.0029**		0.0027
	.1	(0.0012)	(0.0012)		(0.0017)
$OUT(dummy)_t$	$\lambda_{O}^{d}$	, ,	, ,	-0.0032	-0.0067
· · · · · ·	0			(0.0035)	(0.0042)
OUT(log. of $n.inv$ ) <sub>t</sub>	$\lambda_{O}^{log}$			-0.0009	0.0010
	0			(0.0008)	(0.0018)
$INW(dummy)_t$	$\lambda_I^d$			-0.0009	0.0002
( 0)0	1			(0.0030)	(0.0032)
INW(log. of n.inv),	$\lambda_{L}^{log}$			0.0019*	0.0008
				(0.0011)	(0.0015)
$\Delta_{t,t-1}kl$	β	0.2392***		0.2491***	0.2444***
0,0 1	,	(0.0842)		(0.0825)	(0.0850)
$\Delta_{t,t-1}k$	$\beta_k$	· /	$0.1191^{**}$	,	· · · · ·
-,	,		(0.0521)		
$\Delta_{t,t-1}l$	$\beta_l$		$0.3348^{*}$		
			(0.1754)		
$\Delta_{t,t-1}$ HCAP	$\delta_{HCAP}$	0.0003	-0.0013	0.0004	-0.0005
		(0.0137)	(0.0138)	(0.0138)	(0.0137)
$\Delta_{t,t-1}$ HHI	$\delta_{HHI}$	$0.1577^{***}$	$0.1776^{***}$	$0.1666^{**}$	0.1519**
		(0.0740)	(0.0675)	(0.0737)	(0.0730)
$\Delta_{t,t-1}$ INNOV	$\delta_{INNOV}$	0.0008	-0.0010	-0.0002	0.0001
		(0.0100)	(0.0098)	(0.0104)	(0.0098)
Constant	$\alpha$	$0.0270^{***}$	$0.0308^{***}$	$0.0288^{***}$	$0.0293^{***}$
		(0.0039)	(0.0038)	(0.0039)	(0.0039)
Country dummies	$\eta_j$	Yes	Yes	Yes	Yes
Year dummies	$ au_t$	Yes	Yes	Yes	Yes
Industrial mix	$\delta_{SH_{s*}}$	Yes	Yes	Yes	Yes
Observations		746	746	746	746
Regions		255	255	255	255
Significance levels: * 10	$0\%, \frac{**5\%}{5\%}, \frac{***}{5\%}$	1%			
~					

Table 5: Robustness check: specification of the production function and the effect of foreign investments (OLS) 

\_

more productive regions can be differentiated even within a country (different intensities of spatial interactions). In the presence of spatial dependence the inference based on OLS estimates of the Specification (3) may not be reliable. This can be further complicated by the fact that FDI also display a tendency to cluster (as shown in 1) which ends up creating spatial dependence(Blonigen, Davies, Waddell, and Naughton, 2007). In other words, since both productivity and FDI tend to be correlated in space, not accounting for spatial dependence may induces us to wrongly infer a causal relation between FDI and productivity.

Following Elhorst (2010), regional interactions can be modeled using both a spatial autoregressive (or spatial lag) model and a spatial error model. The former assumes that the productivity growth of each region is influenced by that of the neighboring regions. The differenced Equation 5 can be rewritten in the following way, in order to account for spatial interactions in the dependent variable:

$$\Delta y_{ij,t} = \alpha + \lambda W \Delta y_{ij,t} + \beta \Delta k l_{ij,t} + \Delta \mathbf{x}_{ij,t} \delta + + \gamma_O^d OFDI(d)_{ij,t-1} + \gamma_O^{log} OFDI(d)_{ij,t-1} \cdot OFDI(log)_{ij,t-1} + + \gamma_I^d IFDI(d)_{ij,t-1} + \gamma_I^{log} IFDI(d)_{ij,t-1} \cdot IFDI(log)_{ij,t-1} + + \eta_j + \tau_t + \Delta \epsilon_{ij,t} \quad (10)$$

where W represents the spatial weight matrix,  $W\Delta y_{ij,t}$  is the spatially lagged dependent variable,  $\lambda$  is the spatial autoregressive coefficient. In this work we adopt a binary contiguity matrix, in which each  $w_{ij}$  take value '1' or '0', if regions *i* and *j* are, respectively, neighbors or not: we define as neighbors all the regions within a 392 km radius of the region centroid<sup>20</sup>.

A different specification of the spatial dependence is the spatial error model, which posits that, conditional on regressors, the error terms are correlated in space. In our case, the spatial error model can be written as

$$\Delta y_{ij,t} = \alpha + +\beta \Delta k l_{ij,t} + \Delta \mathbf{x_{ij,t}} \delta + + \gamma_O^d OFDI(d)_{ij,t-1} + \gamma_O^{log} OFDI(d)_{ij,t-1} \cdot OFDI(log)_{ij,t-1} + + \gamma_I^d IFDI(d)_{ij,t-1} + \gamma_I^{log} IFDI(d)_{ij,t-1} \cdot IFDI(log)_{ij,t-1} + + \eta_i + \tau_t + \rho W \Delta u_{ij,t} + \Delta \epsilon_{ij,t}$$
(11)

where W represents the spatial weight matrix,  $\Delta u_{ij,t}$  reflects the spatially autocorrelated error term, and  $\rho$  is the spatial autocorrelation coefficient.

Both the spatial lag and spatial error model can be estimated by Maximum Likelihood (ML).

The main difference between the two models is that, in the spatial-lag case, productivity growth of neighboring regions is the channel trough which externalities are transmitted in space, while in the spatialerror model one assumes that the regional dependence arises from the spatial propagation of idiosyncratic shocks. Since we do not have an *a priori* on the shape of regional interactions, we estimate both Equation 10 and 11 by ML, using the the routine developed by Millo and Piras (2009) for the environment R and applying the spatial contiguity matrix previously defined. Results of the estimation are reported in Table 6. Since both the theory and the routine have been defined for balanced panel data, we have to drop some observations in order to balance our panel dataset: the final sample consists of 702 observations and 234 regions. In the first column of Table 6, we report the baseline model –which does not account for spatial interactions- estimated by OLS on the balanced panel (Specification 3\_bal). It is possible to compare it with Specification (3) in Table 3, noting that all the coefficients of the FDI variables slightly drop, both in absolute values and in their statistical significance, due to the sample selection. However, the positive effects of inward and outward FDIs (as well as the threshold effects) are basically unchanged. Estimating the spatial lag model (Specification (3\_splag)) we obtain a spatial autoregressive coefficient ( $\lambda$ ) equal to 0.68, supporting the existence of significant spatial dependence. Nonetheless, all the FDI variables remain significant, and the magnitude of the coefficients does not change much.

 $<sup>^{20}</sup>$ This threshold have been computed as the minimum distance that allow each region to have at least one neighbor, i.e. at least one out-of-diagonal element is equal to one. However, taking a larger radius does not affect the results.

				Specification		
		3_bal	3-splag1	3_splag2	3_sperr1	3_sperr2
Variable	Coefficient					
$OUT_{t-1}(dumy)$	$\mathcal{J}_{p}^{Q}$	$0.0058^{**}$	$0.0043^{*}$	$0.0066^{**}$	$0.0045^{*}$	$0.0051^{*}$
× 1	)	(0.0027)	(0.0024)	(0.0026)	(0.0023)	(0.0026)
$OUT_{t-1}(log. of n.inv)$	$\gamma^{log}_O$	$-0.0021^{*}$	$-0.0018^{*}$	$-0.0043^{***}$	$-0.0020^{**}$	$-0.0040^{***}$
	)	(0.000)	(0.000)	(0.000)	(0.000)	(0.0009)
$\operatorname{IFDI}_{t-1}(\operatorname{dummy})$	$\lambda_p^I$	-0.0057*	$-0.0053^{**}$	-0.0045	$-0.0049^{*}$	-0.0041
	· .	(0.0026)	(0.0026)	(0.0029)	(0.0026)	(0.0029)
$IFDI_{t-1}(log. of n.inv)$	$\gamma_I^{log}$	$0.0021^{*}$	$0.0020^{*}$	$0.0047^{***}$	$0.0023^{**}$	$0.0046^{***}$
	1	(0.0011)	(0.0011)	(0.0011)	(0.0011)	(0.0011)
$\Delta_{t,t-1}kl$	β	$0.2070^{***}$	$0.2372^{***}$	$0.0839^{**}$	$0.2208^{***}$	$0.1624^{***}$
		(0.0782)	(0.0477)	(0.0389)	(0.0460)	(0.0444)
$\Delta_{t,t-1}\mathrm{hcap}$	$\delta_{hcap}$	0.0056	0.0039	0.0182	-0.0010	0.0163
		(0.0135)	(0.0129)	(0.0142)	(0.0124)	(0.0141)
$\Delta_{t,t-1}$ hhi	$\delta_{hhi}$	-0.0178	0.0750	0.0592	0.0814	0.0701
		(0.1105)	(0.0640)	(0.0706)	(0.0630)	(0.0725)
$\Delta_{t,t-1} { m tech}$	$\delta_{tech}$	0.0057	0.0115	$0.0191^{**}$	0.0106	$0.0224^{***}$
		(0.0150)	0.0077)	(0.0081)	(0.0073)	(0.0083)
Constant	σ	$0.0231^{***}$	-0.0031	-0.0005	$0.0219^{***}$	$0.0175^{*}$
		(0.0060)	(0.0055)	(0.0032)	(0.0082)	(0.0091)
(Spatial autoregressive coefficient)	Y		$0.6786^{***}$	$0.7807^{***}$		
			(0.0467)	(0.0357)		
(Spatial autocorrelation coefficient)	θ				$0.7556^{***}$	$0.8137^{***}$
					(0.0406)	(0.0332)
Country dummies		Yes	Yes	No	Yes	No
Year dummies		$\mathbf{Yes}$	$\mathbf{Yes}$	$\mathbf{Yes}$	$\mathbf{Yes}$	$\mathbf{Yes}$
Industrial mix		$\mathbf{Yes}$	$\mathbf{Yes}$	$\mathbf{Yes}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Yes}$
Observations		702	702	702	702	702
Regions		234	234	234	234	234
Significance levels: $* 10\%$ , $** 5\%$ , $***$	1%					

Table 6: Spatial lag and spatial error models (ML)

Specification (3\_splag2) reports the estimation of the spatial lag model without the inclusion of the country dummies: interestingly enough, the specification with the country dummies should be preferred: first, a non-negligible number of country dummies (5 over 19) are significant in Specification (3\_splag) and the null hypothesis that they are jointly significant cannot be rejected; second, the model without country dummies (Specification 3\_splag2) shows an unplausible coefficient of the capital-labour ratio (0.08); third the spatial autoregressive coefficient is larger in the model without the country dummies (0.78), thus indicating their ability in capturing state-specific spatial dependence. This suggests that country dummies may be capturing time-invariant and country-specific spatial specificities, such as institutional characteristics, which cannot entirely be captured by the spatial autoregressive term.

The results of the spatial error model, confirm the presence of spatial dependence, which is indicated by the high and significant spatial autocorrelation coefficient,  $\hat{\rho}=(0.75)$ . In line with the spatial lag model, the coefficient of the dummy variable related to outward investments shrinks with respect to Specification (3\_bal)— from (0.0058) to (0.0045)— and the same is true for the coefficient of the dummy variable of the inward investments —from (0.0057) to (0.0049). This result can be explained by the fact that in the spatial error model, the spatial parameter could pick up the well-known geographical agglomeration phenomenon of the inward foreign investments. The spatial error model has also been estimated without the country dummies, and the results are reported in the last column of the Table. As for the spatial lag model, results support the use of a richer model, including country dummies. In sum, we have showed that even if country dummies capture substantial spatial dependence, some still remains in the residual. However, controlling for such spatial effects does not affect our results, neither in terms of magnitude, nor in terms of statistical significance.

## 6.2.3. Testing for endogeneity

This paper is ultimately interested in estimating the effect of foreign investments flows on productivity growth. However, the relation may also go the other way around: regions with higher productivity growth rates may attract a larger number of foreign investments or may be home to a larger number of investments. In order to test for this possible reverse causality, we use an instrumental variable approach. In particular, we can exploit the fact that foreign investments variables are correlated with past productivity levels and with other regional characteristics at t-1, such as the human capital  $(hcap_{t-1})$ , technology capital  $(tech_{t-1})$ and the degree of sectoral concentration/diversification  $(hhi_{t-1})$ . These variables are good candidates as instruments since they are most likely exogenous to current shocks to productivity growth. GMM estimates are reported in the second column of Table 7 and support our choice of instruments. The Kleibergen-Paap LM underidentification statistics rejects the null hypothesis, supporting our prior that the instruments are in fact correlated with foreign investments. The Hansen-Sargan test does allow to reject the null hypothesis of no overidentification, comforting about the exogeneity of the instruments. Coefficients of inward investments are in line with Specification (3), but with higher standard errors, due to the lower precision of the IV estimates with respect to OLS; coefficients of outward investments are even larger (in magnitude) than those in Specification (3), yielding an higher threshold number of outward investments up to which the effect is positive. However, the endogeneity C test (last row of the Table) does not reject the null hypothesis that FDIs are exogenous, thus OLS should be preferred to the GMM approach in this case.

In the third and fourth column of the Table, we specify a dynamic model: indeed, if higher past productivity growth rates were the cause for an higher number of inward and outward investments, and there was a persistence in regional growth rates we may have captured a spurious correlation between investment flows and current productivity growth, instead of a true 'effect' from FDIs to productivity growth. In the third column (3-gmm2), we test for the endogeneity of the lagged growth rate. Both the under- and over-identification tests confirm the validity of the instruments. Coefficient of both inward and outward investments are stable and significant, and the lagged growth rate is not significant: moreover, the endogeneity test does not reject his null hypothesis. Finally, in the fourth column (3-gmm3) we treat both the lagged growth rate and foreign investments variables as endogenous variables: the tests still suggest that OLS is to be preferred<sup>21</sup>.

 $<sup>^{21}</sup>$ Interestingly enough, excluding the country dummies endogeneity tests would change dramatically. This suggests that

Overall, our tests suggest that, once controlled for country-specific effects, foreign investments are not endogenous to regional productivity growth; thus OLS estimates should be preferred to GMM.

			Specificati	on	
	_	3	3_gmm1	3_gmm2	3_gmm3
Variable	Coefficient				
$OUT_{t-1}(dummy)$	$\gamma_O^d$	$0.0075^{**}$	0.0160	$0.0074^{**}$	0.0351
		(0.0029)	(0.0144)	(0.0030)	(0.0269)
$OUT_{t-1}(log. of n.inv)$	$\gamma_O^{log}$	-0.0029***	-0.0051	-0.0023**	-0.0111
	.0	(0.0010)	(0.0041)	(0.0011)	(0.0077)
$IFDI_{t-1}(dummy)$	$\gamma_I^d$	-0.0072***	-0.0074	-0.0060**	-0.0176
		(0.0027)	(0.0155)	(0.0029)	(0.0351)
$IFDI_{t-1}(log. of n.inv)$	$\gamma_I^{log}$	0.0031***	0.0027	0.0019	0.0089
, ,	1	(0.0012)	(0.0044)	(0.0017)	(0.0066)
$\Delta_{t,t-1}kl$	β	0.2392***	0.2534***	0.2114**	0.2736**
-,		(0.0842)	(0.0953)	(0.0924)	(0.1374)
$\Delta_{t,t-1}$ hcap	$\delta_{hcap}$	0.0003	-0.0003	-0.0026	0.0050
-		(0.0137)	(0.0135)	(0.0138)	(0.0184)
$\Delta_{t,t-1}$ hhi	$\delta_{hhi}$	$0.1577^{**}$	0.1092	0.1201	0.1410
,		(0.0740)	(0.0729)	(0.0755)	(0.0862)
$\Delta_{t,t-1}$ tech	$\delta_{tech}$	0.0008	0.0037	0.0047	-0.0030
		(0.0100)	(0.0100)	(0.0131)	(0.0166)
$\Delta y_{ij,t}$				0.1082	-0.2314
				(0.2578)	(0.4622)
Constant	$\alpha$	0.0270***	0.0244***	0.0235**	0.0263
		(0.0039)	(0.0082)	(0.0092)	(0.0230)
Country dummies	$\eta_j$	Yes	Yes	Yes	Yes
Year dummies	$ au_t$	Yes	Yes	Yes	Yes
Industrial mix	$\delta_{SH_{s*}}$	Yes	Yes	Yes	Yes
Observations		746	746	746	Yes
Regions		255	255	255	255
Underidentification test	(Kleibergen-Pa	ap LM statistic)	20.309	9.838	5.329
Underidentification test	(p-value)		0.0265	0.0432	0.0696
Overidentification test	(Hansen J stati	stic)	4.867	1.181	0.036
Overidentification test	(p-value)		0.8457	0.7576	0.8501
Endogeneity $C$ test			2.473	0.074	1.869
Endogeneity $C$ test (p-	value)		0.6495	0.7853	0.8670
Significance levels: * 10	0%, ** 5%,***	1%			
Cluster-robust standard	l errors in brac	kets			

Table 7: Testing for endogeneity (GMM)

#### 6.2.4. Regional controls in levels

Given the relevance of the country effects in Specification (3), we would like to exclude that our results are biased due to further unobserved regional effects correlated to productivity trends at the regional level. In order to cope with this problem, we can augment Specification (3) by including the set of regional controls in levels at the beginning of the period for each cross-section<sup>22</sup> Moreover, we include the level of labour productivity at the beginning of the period, given that it could explain a significant part of the productivity growth rate (catching-up).

country effects capture unobserved characteristics common to all regions in a country, which are correlated both with FDIs and productivity growth and which, if not taken into account, would bring to a correlation between regressors and the error term due to omitted variables. Results obtained excluding country dummies from the regressors are available from the authors upon requests.

 $<sup>^{22}</sup>$ In principle, one could add regional fixed effects to the equation in first-differences but, one the one hand there is not clear theoretical motive to assume region-specific trends in productivity and, on the other hand, given the short time series, that would leave very little variation to identify our coefficients.

Thus, the vector  $\mathbf{x}_{ij,t-1}$  of regional controls at the beginning of the period can be written as

$$\mathbf{x_{ij,t-1}} = (y_{ij,t-1}, kl_{ij,t-1}, hcap_{ij,t-1}, hhi_{ij,t-1}, tech_{ij,t-1}).$$
(12)

We further include in Equation 6 a vector of time invariant characteristics,  $\mathbf{z}_{ij}\varphi$ , which contains the following information:

- Two dummy variables for coastal (COAST) and capital (CAPT) regions, which take value '1', respectively, in the case in which the region lies on the coast or if it is the capital region of the country. The coastal dummy (information come from Salz, Buisman, Smit, and de Vos, 2006) should account for the general accessibility of a region, which should correlated with its productivity and the degree of internationalization, while the capital dummy is intended to capture agglomeration economies, which could certainly be a driver of productivity growth and which are generally associated with the economic activity and related services taking place in a country's capital.
- We also control for regions which are eligible for European structural funds. A dummy which takes value '1' has been included, when the region is indicated by the European Commission as eligible for 'Objective 1' funds<sup>23</sup>.

# Results are reported in Table 8.

Overall, the effects of inward and outward foreign direct investments on regional productivity are robust both after taking into account the set of regional characteristics at the beginning of the period and the set of time-invariant regional characteristics. Specification (3\_reg3) which is the more demanding, given the high number of covariates and their correlation, shows that the coefficient of the variables related to outward FDIs are still significant even if they slightly decrease in magnitude. Results on the inward FDIs variables are also robust: both the dummy and the continuous variable are significant and they do not change much in terms of magnitude with respect to Specification (3). The capital-labour ratio is stable across all different specifications, while the productivity level at the beginning of the period is never significant, even when it is included without regional controls, as in Specification (3\_reg4).

## 6.2.5. Accounting for the size of regions

To avoid that variables measuring foreign investments capture a generic effect of the 'size' of the region, given that these are the only non-normalized variable on the right-hand side of Equation 6, we included two proxies for the size of regions. In the second column of Table 9 we included a measure of the total population in the vector of regional controls, while in the third column we included the gross value added. Finally, in the fourth column we have normalized the number of investments by the gross value added. Overall, results are not sensitive to the inclusion of a measure of regional size: total population and the gross value added of the region are not significant in Specification (3\_size1) nor in (3\_size2), while the results from Specification (3\_size3), where the number of FDI projects is normalized by the regional value added are in line with Specification (3).

# 7. Concluding remarks

Despite the increasing evidence of integration of sub-national economies in the global arena, and the positive role of multinational firms for economic prosperity in local economies documented in a number of recent studies, evidence on the relationship between foreign investments and regional performance is lacking. Exploiting an original and extensive dataset on FDIs, we investigate the relationship between FDIs and productivity growth in a sample of European regions. The results of the econometric analysis support that both inward and outward foreign direct investments have positive effects on productivity growth at the

 $<sup>^{23}</sup>$ The list of the eligible regions can be found at

http://ec.europa.eu/regional\_policy/objective1/index\_en.htm.

				Specification	L	
Variable	Coefficient	3	$3\_reg1$	$3\_reg2$	3_reg3	$3\_reg4$
$OUT(dummy)_{t-1}$	$\gamma_O^d$	$0.0075^{**}$	$0.0066^{**}$	$0.0064^{**}$	$0.0065^{**}$	$0.0075^{**}$
		(0.0029)	(0.0030)	(0.0031)	(0.0031)	(0.0029)
OUT(log. of n.inv) <sub><math>t-1</math></sub>	$\gamma_O^{log}$	-0.0029***	$-0.0021^{*}$	-0.0022*	-0.0023*	-0.0032***
		(0.0010)	(0.0011)	(0.0011)	(0.0012)	(0.0010)
$INW(dummy)_{t-1}$	$\gamma_I^d$	-0.0072***	-0.0067**	-0.0068**	-0.0068**	-0.0067**
		(0.0027)	(0.0028)	(0.0028)	(0.0028)	(0.0027)
INW(log. of $n.inv$ ) <sub>t-1</sub>	$\gamma_I^{log}$	$0.0031^{***}$	$0.0023^{*}$	$0.0022^{*}$	$0.0022^{*}$	$0.0027^{**}$
	1	(0.0012)	(0.0012)	(0.0012)	(0.0012)	(0.0012)
$\Delta_{t,t-1}kl$	β	0.2392***	0.2620***	0.2647***	0.2601***	0.2345***
		(0.0842)	(0.0970)	(0.0986)	(0.0996)	(0.0825)
$\Delta_{t,t-1}$ hcap	$\delta_{hcap}$	0.0003	0.0020	0.0016	0.0019	-0.0002
	-	(0.0137)	(0.0140)	(0.0140)	(0.0141)	(0.0138)
$\Delta_{t,t-1}$ hhi	$\delta_{hhi}$	$0.1577^{**}$	0.1181	0.1195	0.1181	$0.1464^{*}$
		(0.0740)	(0.0775)	(0.0775)	(0.0778)	(0.0758)
$\Delta_{t,t-1}$ tech	$\delta_{tech}$	0.0008	0.0003	0.0014	0.0013	0.0005
		(0.0100)	(0.0107)	(0.0108)	(0.0108)	(0.0100)
$y_{t-1}$	$\phi_y$				0.0062	0.0093
					(0.0131)	(0.0077)
$kl_{t-1}$	$\phi_{kl,t-1}$		0.0069	0.0070	0.0054	
			(0.0062)	(0.0062)	(0.0072)	
$hcap_{t-1}$	$\phi_{hcap}$		0.0033	0.0031	0.0038	
			(0.0056)	(0.0058)	(0.0060)	
$hhi_{t-1}$	$\phi_{hhi}$		-0.0379**	-0.0377**	-0.0372**	
			(0.0184)	(0.0182)	(0.0183)	
$\operatorname{tech}_{t-1}$	$\phi_{tech}$		0.0017	0.0023	0.0021	
			(0.0016)	(0.0016)	(0.0017)	
COAST	$\varphi_{COAST}$			0.0024	0.0023	
				(0.0017)	(0.0017)	
CAPT	$\varphi_{CAPT}$			0.0038	0.0035	
				(0.0033)	(0.0034)	
OBJ1	$\varphi_{OBJ1}$			0.0031	0.0035	
				(0.0026)	(0.0026)	
Constant	$\alpha$	$0.0270^{***}$	-0.0597	-0.0536	-0.0599	-0.0084
		(0.0039)	(0.0376)	(0.0378)	(0.0387)	(0.0296)
Country dummies	$\eta_j$	Yes	Yes	Yes	Yes	Yes
Year dummies	$ au_t$	Yes	Yes	Yes	Yes	Yes
Industrial mix <sup>*</sup>		Yes	Yes	Yes	Yes	Yes
Observations		746	746	746	746	746
Regions		255	255	255	255	255

Table 8: Robustness check: regional characteristics (OLS)

\* The industrial mix include both  $\Delta_{t,t-1}SH_{s*}$  (differences) and  $SH_{s*ji,t-1}$  (lagged) Significance levels: \* 10%, \*\* 5%,\*\*\* 1% Cluster-robust standard errors in brackets

			Specifi	cation	
Variable	Coefficient	3	3_size1	3_size2	3_size3
$OUT(dummy)_{t-1}$	$\gamma_O^d$	$0.0075^{**}$	$0.0079^{***}$	$0.0079^{***}$	$0.0074^{***}$
	-	(0.0029)	(0.0030)	(0.0030)	(0.0026)
OUT(log. of n.inv) $_{t-1}$	$\gamma_{O}^{log}$	-0.0029***	-0.0025***	-0.0024**	
	10	(0.0010)	(0.0009)	(0.0010)	
$INW(dummy)_{t-1}$	$\gamma_I^d$	-0.0072***	-0.0067**	-0.0069***	-0.0073***
	1	(0.0027)	(0.0026)	(0.0026)	(0.0025)
INW(log. of $n_{inv}$ ) <sub>t-1</sub>	$\gamma_{I}^{log}$	0.0031***	0.0034***	0.0035***	· /
	11	(0.0012)	(0.0012)	(0.0012)	
OUT(log_of_n inv/gva),	$\gamma^{log}$	()	()	()	-0.0025**
0 0 1 (108) 01 11111 ( 8 0) 1-1	10				(0.0010)
INW(log of n inv/gva).	$\sim^{log}$				0.0036***
$\Pi(\Pi(\log, \log, \log, \min(\log, \alpha))) = 1$	/1				(0.0000)
$\Delta_{++-1}kl$	в	0 2392***	0 2351***	0.2376***	0.2372***
= t, t = 1	Ρ	(0.0842)	(0.0848)	(0.0849)	(0.0844)
Att 1HCAP	<b><i>SHCAR</i></b>	0.0003	-0.0002	-0.0001	0.0001
<u> </u>	SHCAF	(0.0137)	(0.0137)	(0.0138)	(0.0137)
$\Delta_{t,t-1}$ HHI	$\delta_{HHI}$	0.1577**	0.1555**	0.1575**	0.1577**
	- 11 11 1	(0.0740)	(0.0737)	(0.0739)	(0.0740)
$\Delta_{t,t-1}$ INNOV	διννον	0.0008	0.0008	0.0007	0.0007
<i>t,t</i> <b>1</b>	1111107	(0.0100)	(0.0099)	(0.0099)	(0.0099)
Constant	$\alpha$	0.0270***	0.0491***	0.0146	0.0363***
		(0.0039)	(0.0155)	(0.0130)	(0.0103)
$pop_{t-1}$	$\phi_{non}$		-0.0018		
	/ F = F		(0.0012)		
value $added_{t-1}$	$\phi_{ava}$		. ,	-0.0017	
	. 5			(0.0015)	
Country dummies	$\eta_j$	Yes	Yes	Yes	Yes
Year dummies	$ au_t$	Yes	Yes	Yes	Yes
Industrial mix <sup>*</sup>		Yes	Yes	Yes	Yes
Observations		746	746	746	746
Regions		255	255	255	255
* [] . ] ] ]	1 1 4	CII (1.C		(1	1)

Table 9: Robustness check: regional size (OLS)  $\,$ 

\* The industrial mix include both  $\Delta_{t,t-1}SH_{s*}$  (differences) and  $SH_{s*ji,t-1}$  (lagged) Significance levels: \* 10%, \*\* 5%,\*\*\* 1% Cluster-robust standard errors in brackets

regional level, after controlling for a relevant set of regional characteristics, such as human and technological capital, industry mix, and productivity trends at the country level. The econometric analysis has provided –to our knowledge for the first time– a robust evidence of positive effects in a large set of NUTS2 regions in almost all countries of the European Union (EU-27). Previous studies with a regional perspective have focused on comparisons within single countries and have addressed only the role of 'inward' investments as a driver of increasing local performance. Moreover, those few studies which have attempted to assess the specific role of outward investments on productivity have taken a country perspective, almost neglecting the sub-national level of analysis. This is most unfortunate, given that the regional level of analysis is particularly appropriate to capture indirect and compositional effects of FDIs. Our results are consistent with the idea that direct effects of MNEs on productivity and positive indirect effects (i.e. pecuniary and technology externalities) prevail over negative indirect effects (crowding-out and business stealing effects), thus resulting in a positive effect on aggregate productivity. This is in line with previous empirical literature on the entry of MNEs, finding a positive direct contribution to the productivity of the host economy; moreover, it reinforces the (scatter) previous evidence on the positive effects of having a larger number of outward investing firms in a territory.

Our specification allows to add an important qualification to previous results. In particular, inward foreign investments have a positive effect on regional productivity only above a certain threshold level. This result can be explained by the fact that, even large firms, such as multinationals, produce a relatively small value added in the host country with respect to the economy of a NUTS2 region. Therefore, entry of one or few multinationals make a relatively small contribution to the aggregate productivity, and it requires several foreign entries, to make a appreciable direct effect. On the other hand outward investments seem to have a positive effects up to a certain threshold, which is however very high in our sample. Results from our preferred specification suggest that about 30% of regions have higher productivity growth, thanks to the relatively large flows of inward investments, while in 50% of the cases productivity growth is higher due to outward investments.

These results are robust to different specifications of the econometric model, like the inclusion of a number of regional characteristics, controls for spatial dependence in productivity growth across European regions, and for the endogeneity of FDIs. The positive effects of inward and outward FDIs are robust and quite stable also in terms of magnitude of the coefficients. Admittedly, the effect is rather small in magnitude, but it should be noted that the effect of other important regional characteristics such as human and technological capital, regional size, being a coastal region, hosting the country's capital and being eligible for Objective 1 Structural Funds is not significantly different from zero.

In conclusion, our results support that both inward and outward FDIs can bring significant benefits to regional economies by increasing productivity growth. This has important implications for local and national policy. One the one hand, governments should implement policies to attract inward FDIs conducive to higher productivity growth, but the effort must be substantial, so that foreign entries reach the threshold level required to determine positive effects. On the other hand, the fear of hollowing-out European knowledge which has accompanied measures aimed at reducing outward investments is not completely founded. Our results suggest that up to a certain point it is good for a region that local firms invest abroad. Thus, this calls for policies aimed at removing the obstacles to foreign investments.<sup>24</sup>

 $<sup>^{24}</sup>$ Admittedly, many policies limiting outward investments were also motivated by the fear of job losses. While we cannot say anything on the effect on regional employment here, we argue that higher productivity growth is likely to increase jobs in the medium-run, whatever the displacement effect in the short run.

# A. Appendix

#### A.1. Labour productivity

Some remarks on the labour-productivity measure should be made. First, data on the regional employment are drawn from the European Regional Database. We chose to use this source, since the employment series of the Regio database has a higher number of missing values which would have decreased the set of regions under analysis. The downside of this choice is that in the version of the European Regional Database available to us, values for 2005 and 2006 were forecast. However, we checked that correlation with the actual (non missing) values, reported by the more updated Regio dataset is very high (0.95). Second, in order to build deflators for regions belonging to Cyprus, Estonia, Latvia, Lithuania and Malta (which are actually all single-region country) we have used the series of price index in the previous release of the EU KLEMS database (2008) given that they were not available in the last release yet. Third, for Bulgarian and Romanian regions we have used the 'Eurozone' series of price index, given that the national series were not available in the database.

In Tables 10 and 11 we show that regions belonging to EU-12 New Member States show (on average) an higher labour productivity growth rate (5%) with respect to regions belonging to 'Old' EU-15 countries (1.2%). This is in line with the literature that claims for the role of the economy restructuring and catchingup to the technological frontier as the main explanations for this phenomenon. Among the countries in the EU-15, it is possible to appreciate a certain amount of heterogeneity in growth rates. United Kingdom, Italy, Spain and Portugal show low performance in terms of labour productivity growth during the period 2003-2006. France and Germany show modest growth trends. Ireland shows the best performance on average, even showing a large standard error, which is likely due to the big difference between the region of Dublin (IE02) which saw a strong economic performance over the past number of years, and the other region (Border, Midland and Western, IE01); some North-European countries show fast growth rates, as the Netherlands (2.7%), Sweden (2.6%), and Finland (2.5%), which is in line with previous analysis at the country level (see O'Mahony, Rincon-Aznar, and Robinson, 2010, among others). Among the New Member States, Romania, Slovakia, Lithuania and Czech Republic show the best performance in terms of labour productivity growth. It is interesting to note the relative higher standard deviations in the growth rates of regions belonging to EU-12 with respect to regions belonging to the 'Old' member states. This is probably due to the fact that there is a considerable amount diversity in growth experience: for example in Romania, the capital region (RO32) shows the highest growth rate (0.169), while other regions perform differently (RO12, RO21, RO22); in the Czech Republic, Moravskoslezsko (CZ08) — which benefits from its location on the borders of Poland and Slovakia —, the Central Bohemian Region (CZ02) and the region of Prague (CZ01) show the best performance in terms of labour productivity growth, while the North East (CZ05) performs rather poorly.

#### A.2. Capital-labour ratio

We have included the capital-labour ratio  $(KL_{ijt})$  in Equation 1, in order to control for the regional factor share. The variable has been computed as the ratio of the regional capital stock  $(K_{ijt})$  to employment (thousands) in the region  $(L_{ijt})$ . The capital stock at the regional level, has been obtained applying the perpetual inventory method (PIM) to the series of capital investments in the region (at 1995 prices in millions of euro)<sup>25</sup> taken from the European Regional Database. As for the employment series, capital investments' information for 2005 and 2006 are forecast.

We followed Hall and Mairesse (1995), and the capital stock at the beginning of the first year has been defined as below:

$$K_{ij,t=1} = \frac{I_{ij,t=1}}{g_{ij} + \delta},$$
 (13)

 $<sup>^{25}</sup>$ The series comprehend aggregate investments by the following sectors: agriculture, total energy and manufacturing, construction, market and non-market services.

$ \begin{array}{llllllllllllllllllllllllllllllllllll$
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$
8         8         8         8         8         8         8         8         8         8         8         10000         1000         1000
Italy 0.005 0.016 0.056 0.008 0.019 0.001 0.000 -0.004 0.008 -0.002
(0.019)  (0.011)  (0.060)  (0.029)  (0.084)  (0.009)  (0.006)  (0.010)  (0.020)  (0.015)
84         84         84         80         84         80         80         80         80         80         80
Luxembourg 0.026 0.022 0.057 0.018 0.051 -0.003 0.001 -0.005 0.013 -0.005
(0.022)  (0.003)  (0.329)  (0.017)  (0.039)  (0.003)  (0.002)  (0.003)  (0.011)  (0.009)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$
Netherlands 0.027 0.034 0.049 0.008 0.006 0.000 -0.004 0.002 0.003 0.000
(0.019)  (0.030)  (0.056)  (0.029)  (0.063)  (0.007)  (0.006)  (0.008)  (0.013)  (0.014)
48 $48$ $48$ $44$ $48$ $44$ $44$ $44$
Portugal 0.010 0.029 0.070 0.007 0.130 -0.004 0.000 -0.004 0.008 0.002
(0.015)  (0.024)  (0.125)  (0.016)  (0.192)  (0.005)  (0.005)  (0.009)  (0.014)  (0.010)
20 $20$ $20$ $15$ $20$ $15$ $15$ $15$ $15$ $15$ $15$
Sweden 0.026 0.017 0.034 0.006 -0.040 0.002 -0.002 -0.003 0.001 0.002
(0.024) $(0.023)$ $(0.020)$ $(0.024)$ $(0.076)$ $(0.004)$ $(0.004)$ $(0.007)$ $(0.011)$ $(0.009)$
32 $32$ $32$ $32$ $32$ $32$ $32$ $32$
United Kingdom -0.004 0.019 0.032 0.014 -0.003 0.002 -0.003 -0.004 0.006 0.000
(0.051) $(0.021)$ $(0.059)$ $(0.033)$ $(0.088)$ $(0.012)$ $(0.009)$ $(0.011)$ $(0.018)$ $(0.019)$
144 $144$ $136$ $121$ $144$ $121$ $121$ $121$ $121$ $121$ $121$ $121$
EU_15 0.012 0.021 0.034 0.011 0.025 0.000 -0.002 -0.003 0.005 0.000
(0.028) $(0.024)$ $(0.074)$ $(0.035)$ $(0.107)$ $(0.010)$ $(0.009)$ $(0.011)$ $(0.016)$ $(0.016)$
816 816 788 709 816 709 709 709 709 709 709
Note: the average is reported in the first row; the standard deviation is reported in brackets in the second row

Table 10: Growth rates by country, EU15, 2003-2006

and the third row shows the number of observations (region/year)

\_

	$\Delta y$	$\Delta kl$	$\Delta h cap$	$\Delta hhi$	$\Delta tech$	$\Delta SH_{EF}$	$\Delta SH_HT$	$\Delta SH_{LT}$	$\Delta SH_KIS$	$\Delta SH_LKIS$
Bulgaria	0.023	-0.006	0.006	-0.001	0.112	0.006	0.001	0.002	-0.001	0.002
	(0.047)	(0.024)	(0.041)	(0.020)	(0.183)	(0.007)	(0.004)	(0.008)	(0.009)	(0.009)
	18	24	18	18	24	18	18	18	18	18
Cyprus	0.030	0.023	0.008	0.005	0.065	0.003	0.000	-0.004	0.005	-0.001
	(0.013)	(0.003)	(0.043)	(0.022)	(0.113)	(0.007)	(0.002)	(0.009)	(0.008)	(0.008)
	3	4	4	4	4	4	4	4	4	4
Czech Republic	0.066	0.003	0.030	0.004	0.088	0.000	0.004	-0.003	0.003	0.000
	(0.054)	(0.018)	(0.050)	(0.029)	(0.116)	(0.009)	(0.007)	(0.009)	(0.008)	(0.012)
	32	32	32	32	32	32	32	32	32	32
Estonia	0.030	0.069	0.029	-0.003	0.061	0.009	-0.001	-0.003	-0.005	0.003
	(0.008)	(0.005)	(0.044)	(0.038)	(0.059)	(0.005)	(0.013)	(0.008)	(0.025)	(0.008)
	3	4	4	4	4	4	4	4	4	4
Hungary	0.012	0.067	0.031	0.012	0.084	0.003	0.000	-0.007	0.003	0.004
	(0.048)	(0.029)	(0.050)	(0.028)	(0.120)	(0.007)	(0.010)	(0.010)	(0.012)	(0.014)
	28	28	28	28	28	28	28	28	28	28
Lithuania	0.073	0.047	0.061	0.009	0.280	0.006	0.000	0.000	0.002	0.008
	(0.004)	(0.013)	(0.018)	(0.007)	(0.262)	(0.008)	(0.003)	(0.003)	(0.006)	(0.002)
	3	4	4	4	4	4	4	4	4	4
Latvia	0.012	0.082	0.030	0.009	0.148	0.008	-0.001	-0.004	0.002	0.005
	(0.051)	(0.008)	(0.073)	(0.020)	(0.168)	(0.005)	(0.003)	(0.007)	(0.008)	(0.005)
	3	4	4	4	4	4	4	4	4	4
Malta	-0.012	0.015	0.060	0.021	0.095	0.002	-0.004	-0.006	0.006	0.004
	(0.042)	(0.010)	(0.072)	(0.027)	(0.110)	(0.009)	(0.015)	(0.008)	(0.005)	(0.014)
	3	4	4	4	4	4	4	4	4	4
Poland	0.034	0.032	0.087	-0.004	0.199	0.004	0.001	-0.002	0.002	0.002
	(0.085)	(0.018)	(0.051)	(0.022)	(0.275)	(0.008)	(0.006)	(0.011)	(0.012)	(0.014)
	64	64	64	32	64	32	32	32	32	32
Romania	0.139	0.016	0.056	-0.016	0.104	0.003	0.000	-0.001	0.003	0.009
	(0.078)	(0.024)	(0.087)	(0.054)	(0.451)	(0.009)	(0.010)	(0.013)	(0.009)	(0.013)
	24	32	32	32	32	32	32	32	32	32
Slovenia	0.026	0.073	0.092	0.011	0.143	0.000	-0.002	-0.007	0.008	0.000
	(0.025)	(0.004)	(0.066)	(0.029)	(0.102)	(0.002)	(0.009)	(0.005)	(0.005)	(0.010)
	4	4	4	4	4	4	4	4	4	4
Slovakia	0.074	0.029	0.067	0.007	0.082	0.003	0.003	-0.004	0.002	0.002
	(0.043)	(0.026)	(0.051)	(0.024)	(0.137)	(0.007)	(0.008)	(0.009)	(0.013)	(0.012)
	16	16	16	16	16	16	16	16	16	16
EU_12	0.050	0.028	0.054	0.001	0.129	0.003	0.001	-0.003	0.002	0.003
	(0.075)	(0.032)	(0.063)	(0.033)	(0.252)	(0.008)	(0.008)	(0.010)	(0.010)	(0.012)
	201	220	214	182	220	182	182	182	182	182
Note: the averag	e is repor	ted in the	first row;	the stand	lard devia	tion is repor	rted in brack	xets in the s	second row	

Table 11: Growth rates by country, EU12, 2003-2006  $\,$ 

and the third row shows the number of observations (region/year)

where  $I_{ij,t=1}$  is the amount of capital investments taken by the region *i* in the first year of the series<sup>26</sup>,  $g_{ij}$  is the rate of growth of capital investments observed in the region in a given span of time (in this case is from 1995-2002<sup>27</sup>), and  $\delta$  is depreciation rate which has been set equal to 7.5%<sup>28</sup>. Capital stock from the second year onward has been computed using the following formula:

$$K_{ij,t} = (1 - \delta) \cdot K_{ij,t-1} + I_{ij,t}.$$
(14)

The variable has been included in logs in the econometric analysis,  $kl_{ijt}$ .

#### A.2.1. Other regional characteristics

In this Section, we detail how regional characteristics — i.e. the level of human capital, the technological capital and the regional industrial mix — have been measured.

- Human capital  $(HCAP_{ijt})$  has been proxied by the (log of the) share of population aged 25 or more (thousands) with tertiary-type education degree (ISCED 5-6) in each region. Information come from the EU Regional Database, maintained by Eurostat.
- The regional technological capital  $(TECH_{ijt})$  has been proxied by the ratio of the stock of patents applications  $(INNOV_{ijt})$  to the total population (thousands) in the region  $(POP_{ijt})$ . The stock has been recovered using information on the number of patent applications to the European Patent Office (EPO) coming from each European region, which are available in the database maintained by Eurostat<sup>29</sup>. Data on total population comes from the database developed by Cambridge Econometrics. The stock for the years t = (2003, 2004, 2005, 2006) has been computed as the sum of the patent applications in all sectors in the previous five years  $(PATAPP_{ijt})$ :

$$INNOV_{ij,t} = \sum_{t=t-5}^{t} PATAPP_{ijt}.$$
(15)

The ratio has been included in logs in the econometric analysis,  $tech_{ijt}$ .

• We have taken into account the regional industrial mix  $(SH_{s*ijt})$ , by introducing the share of employment in six broad sectors  $s^*$  of the regional economy: Agriculture, hunting, forestry and fishing (AC), Electricity, gas, water supply and Constructions (EF), High-tech manufacturing & Medium high-tech manufacturing (HD), Medium low-tech manufacturing & Low-tech Manufacturing (LD), Knowledge-intensive services (KI) and Less knowledge-intensive (LKI) services. Each share has been computed in the following way:

$$SH_{s*ijt} = \frac{L_{s*ijt}}{L_{ijt}}$$

where  $L_{ijt}$  and  $L_{s*ijt}$  denote, respectively, total employment in the region *i* which belongs to country *j* (thousands), and employees belonging to the sector *s*\*. To avoid multicollinearity we introduced five coefficients in the regressions. The excluded sectoral share is the AC sector (Agriculture, hunting, forestry, fishing, mining and quarrying). Data regarding employees in each sector come from the database maintaned by Eurostat.

 $<sup>^{26}</sup>$ We start computing the capital stock series at 1995 up to 2006, even if in the econometric analysis we use the values from 2002 to 2006. The main motivation relates to the possibility to rest on a more reliable capital stock at the left hand side of Equation 14 for the years under analysis.

 $<sup>^{27}</sup>$ For Romanian regions the investments' growth rate has been computed for the period 1998-2002, given the lack of data for the years 1995, 1996 and 1997.

 $<sup>^{28}</sup>$ As robustness checks we also computed the capital stock assuming depreciation rate of 5% and 10%, and we did not register significantly different results.  $^{29}$ Data on patent applications are regionalised on the basis of the investors' residence: in the case of multiple investors

<sup>&</sup>lt;sup>29</sup>Data on patent applications are regionalised on the basis of the investors' residence: in the case of multiple investors proportional quotas have been attributed to each region.

Data on employment by sectors are missing for a number of (region/year) observations; in order not to loose those observations, we have used linear interpolation to fill the gaps for all the observations that were 'missing', but which had 'non-missing' observations the year before and the year after the missing ones. We further filled in a small amount of missing observations in the High-tech manufacturing sector (which showed the highest number of missing observations) as the difference between total regional employment and the sum of employees in all the others sectors (AC, EF, Medium-high tech manufacturing, Medium-low tech manufacturing, Low-tech manufacturing, KI, LKI).

• We have controlled for the degree of concentration/diversification of the regional industrial mix. Following the literature (see Cingano and Schivardi, 2004; Bracalente and Perugini, 2008, among others), we have used the Herfindahl-Hirschman index as a proxy for concentration/diversification computed as follows:

$$HHI_{ijt} = \sum_{s} SH_{sijt}^2 = \sum_{s} \left(\frac{L_{sijt}}{L_{ijt}}\right)^2,\tag{16}$$

where  $SH_{sijt}$  are a more detailed disaggregation of the employment shares defined above. In fact, as elements of the HHI we take into account 8 broad sectors, s: Agriculture, hunting, forestry and fishing (AC), Electricity, gas, water supply and Constructions (EF), High-tech manufacturing (HTD), Medium high-tech manufacturing(MHTD), Medium low-tech manufacturing (MLTD), Low-tech Manufacturing (LTD), Knowledge-intensive services (KI) and Less knowledge-intensive (LKI) services. In particular, we consider the HTD and the MHTD as two separate sectors here, and the same holds for the LTD and the MLTD which are considered separate elements of the  $HHI^{30}$ . The HHI index, which is equal to '1' for regions with all employees in one sector and which goes toward '0' for more diversified regional structures, allows us to control for the sectoral concentration/variety of the region, while by introducing the  $SH_{s^*it}$  ratios, we account for the different 'quality' of the industrial mix. For any given level of HHI we expect regional productivity to be higher in regions where the share of high-value added activities (such as High-tech Manufacturing and Knowledge-intensive services) is higher<sup>31</sup>.

The *HHI* enters in logs in the econometric analysis, *hhi*.

The taxonomy of broad sectors —which have been used in order to build the Herfindahl index of diversification and the shares of employment which proxy the regional industrial mix— has been taken from the list which has been proposed by Eurostat in the EU regional database. We cross-refer the reader to the technical repost by Felix (2006) for further details on the employed taxonomy. Sectors are presented in Table 12.

# A.3. List of regions

The list of the NUTS 2 regions which have been considered in the baseline Specification (3) is reported in Table 13. Overall, we can account for 255 regions (and 746 observations) belonging to the EU in our analysis, for the period 2003-2006.

 $<sup>^{30}\</sup>mathrm{The}$  detailed taxonomy of sectors s is presented in Table 12 of the Appendix

 $<sup>^{31}</sup>$ The use of different levels of aggregation in the HHI with respect to these employments shares is motivated both by the achieved greater precision of the Herfindahl-Hirschman index, which aims at capturing the variability in the regional industrial mix, and –on the contrary– by the attempt to minimize over-specification in the estimates of the coefficients of the sectoral employment shares.

	TADE 12. DEGARDOWIL OF SECONDS (NACE INCV. 1.1 COUCS)
Agriculture, hunting, forestry and fishing Electricity, gas, water supply and constructions	01 to 05 Agriculture, hunting, forestry and fishing 40 to 41; 45 Electricity, gas, water supply and constructions
High-tech Manufacturing	30 Manufacture of office machinery and computers 32 Manufacture of radio, television and communication equipment and apparatus 33 Manufacture of medical, precision and optical instruments, watches and clocks
Medium High-tech Manufacturing	<ul> <li>24 Manufacture of chemicals and chemicals products</li> <li>29 Manufacture of machinery and equipment n.e.c.</li> <li>31 Manufacture of electrical machinery and apparatus n.e.c.</li> <li>34 and 35 Manufacture of transport equipment</li> </ul>
Low and medium-low-tech Manufacturing	15 to 22 Manufacture of food products, beverages and tobacco; textiles and textile products; leather and leather products; wood and wood products; pub, paper and paper products, publishing and printings 23 Manufacture of coke, refined petroleum products and nuclear fuel 25 to 28 Manufacture of rubber and plastic products; basic metals and fabricated metals product; other non-metallic mineral products 36 to 37 Manufacturing n.e.c.
Knowledge-intensive services	<ul> <li>61 Water Transport</li> <li>62 Air Transport</li> <li>64 Post and telecommunications</li> <li>65 to 67 Financial intermediation</li> <li>70 to 74 Real estate, renting and business activities</li> <li>80 Education</li> <li>85 Health and social work</li> <li>92 Recreational, cultural and sporting activities</li> </ul>
Less knowledge-intensive services	<ul> <li>50 to 52 Motor trade</li> <li>55 Hotels and restaurants</li> <li>60 Land transport ; transport via pipelines</li> <li>63 Supporting and auxiliary transport activities, activities of travel agencies</li> <li>75 Public administration and defence; compulsory social security</li> <li>90 Sewage and refuse disposal, sanitation n.e.c.</li> <li>93 Other service activities</li> <li>95 Activities of households as employers of domestic staff</li> <li>99 Extra-territorial organizations and bodies</li> </ul>

1 1 codes) Rev S (Na Table 12. Breakdo

002		TOPA	Contrainidad de Macilio	Liboo		TOATICIA MUMBINA TERM	11.175		COTRE	COLT
2000		1	Aragón	ES24	ozen	Provincia Autonoma Bolzano-B	IIDI	Azur	Provence-Alpes-Cöte d	FR82
			La Rioja	ES23		Lombardia	ITC4		Languedoc-Roussillon	FR81
	Limburg(NL)	NL42	Comunidad Foral de Navarra	ES22		Liguria	ITC3		Auvergne	FR72
	Noord-Brabant	NL41	Pais Vasco	ES21		Valle d'Aosta/Vallée d'Aoste	ITC2		Rhône-Alpes	FR71
	Zeeland	NL34	Cantabria	ES13		Piemonte	ITC1		Limousin	FR63
	Zuid-Holland	NL33	Principado de Asturias	ES12	21		Italy		Midi-Pyrénées	FR62
	Noord-Holland	NL32	Galicia	ES11		Southern and Eastern	IE02		Aquitaine	FR61
	Utrecht	17 NL31	BILLANDC	Spain	ŀ	Border, Midlands and Western	IE01		Poitou-Charentes	FR53
	Flowland	22 TN	Classonia	SIU	c,	Del-Alloid	Incland		r ays ue la Lolle Brotanno	LOLD T
	Calded and	1 NL 21	Vychodne Slovensko	SK04		Eszak-Altold D4LAI614	HU32		Para de la Leire	FR43
	Drentine	INLIG	Stredne Slovensko	SN03		Észak-Magyaroiszag	HU31		Alsace	FR42
	Friesland(NL)	NL12	Západné Slovensko	SK02		Dèl-Dunántůl	HU23		Lorraine	FR41
	Groungen	NLII	Bratislavský	SK01		Nyugat-Dunantul	HU22		Nord-Pas-de-Calais	FR30
12	erlands	4 Neth		Slovakia		Közép-Dunántúl	HU21		Bourgogne	FR26
	0 Northern Ireland	UKN	Vest	RO42		Közép-Magyarotszág	HU10		Basse-Normandie	FR25
	6 Highlands and Islands	UKM	Sud-Vest	RO41	7		Hungary		Centre	FR24
	5 North Eastern Scotland	UKM	Bucuresti	RO32		Kriti	GR43		Haute-Normandie	FR23
	3 South Western Scotland	UKM	Sud	RO31		Notio Aigaio	GR42		Picardie	FR22
	2 Eastern Scotland	UKM	Sud-Est	RO22		Attiki	GR30		Champagne-Ardenne	FR21
	? East Wales	UKL	Nord-Est	RO21		Peloponnisos	GR25		Ile de France	FR10
	West Wales and The Valleys	UKL	Centru	RO12		Sterea Ellada	GR24	22		France
	4 Devon	UKK	Nord-Vest	RO11		Dytiki Makedonia	GR23		Åland	F120
	3 Cornwall and Isles of Scilly	8 UKK		Romania		Ionia Nisia	GR22		Pohjois-Suomi	FIIA
	2 Dorset and Somerset	UKK	Alentejo	PT18		Ipeiros	GR21		Länsi-Suomi	F119
	1 Gloucestershire, Wiltshire and North Somerset	UKK	Lisboa	PT17		Thessalia	GR14		Etelä-Suomi	FI18
	Kent	UKJ	Centro (PT)	PT16		Dytiki Makedonia	GR13		Itā-Suomi	F113
	Hampshire and Isle of Wight	UKJS	Algarve	PT15		Kentriki Makedonia	GR12	57		Finland
	Surrey, East and West Sussex	UKJS	Norte	PT11	-	Anatoliki Makedonia, Thraki	GR11		Estonia	EE00
	Berkshire, Bucks and Oxfordshire	5 UKJ		Portugal	12	c	Greece	1		Estonia
	London	UKI	Pomorskie	PL63		Thüringen	DEG0		Moravskoslezko	CZ08
	Figger	UKH	Warminsko-Mazurskie	PL62		Schleswig-Holstein	DEF0		Strední Morava	C/207
	2	INT	Winwelo-Pomorekia	p161		Sachean-Anhalt	DEEN		liboxéebod	C/206
	1 Foot Angelia	UNG	Onolebio	PL91		Loresden	DED2		Severozapad	C705
	2 Shropshire and Statfordshire	UKG	Deletate	PL43		Chemnitz	DEDI		Jihozapad	CZ03
	1 Heretordshire, Worcestershire and Warks	UKG	Zachodniopomorskie	PL42		Saarland	DECO		Stredni Cechy	C/202
	Lincolnshire	UKF	Wielkopolskie	PL41		Rheinhessen-Pfalz	DEB3		Praha	CZ01
	? Leicestershire, Rutland and Northants	UKE	Podlaskie	PL34		Trier	DEB2	8	G	Czech Republic
	Derbyshire and Nottinghamshire	UKF	Swietokrzyskie	PL33		Koblenz	DEB1		Cyprus	CY00
	4 West Yorkshire	UKE	Podkarpackie	PL32		Arnsberg	DEA5	1		Cyprus
	3 South Yorkshire	UKE	Lubelskie	PL31		Detmold	DEA4		Yuzhen tsentralen	BG42
	North Yorkshire	UKE	Slaskie	PL22		Münster	DEA3		Yugozapaden	BG41
	Fast Riding and North Lincolnshire	UKE	Malonolskie	PL21		Köln	DEA2		Yugoiztochen	BG34
	5 Mersevside	UKD	Mazowieckie	PL12		Disseldorf	DEA1		Severoiztochen	BG.33
	1 Greater		Lydelde	Poland		Woorr Erro	DE93		Severozapaden	BC 31
	Cheshire		Maita	MITOO		Hannover	DE02	0	0	Bulgaria
	1 Cumbria			Malta		Braunschweig	DE91	5	Prov. Namur	BE35
	2 Northumberland, Tyne and Wear	UKC	Luxembourg	L'U00		Mecklenburg-Vorpommern	DE80		Prov. Luxembourg (B)	BE34
	t Tees Valley and Durham	1 UKC		Luxembourg		Kassel	DE73		Prov. Liège	BE33
36	ad Kingdom	Unit	Lithuania	LT00		Gießen	DE72		Prov. Hainaut	BE32
	Övre Norrland	1 SE33		Lithuania		Darmstadt	DE71		Prov. Brabant Wallon	BE31
	Mellersta Norrland	SE32	Latvia	LV00		Hamburg	DE80	-	Prov. West-Vlaanderen	BF25
	V and verige	1 20220	on ugua	Latvia		Bromon	DESO		Prov. Vlaame Brahant	BE94
	Niet avorigo	SE22	Sordamo	ITC?		Brandenburg-Nordost Brandenburg-Sidwost	DE41		Prov. Lunburg (B) Prov. Ocet-Vleenderen	BE22 RE93
	Småland med öarna	SE21	Calabria	ITF6		Berlin	DE30		Prov. Antwerpen	BE21
	Ostra Mellansverige	SE12	Basilicata	ITF5		Schwaben	DE27		Brussels	BE10
	Stockholm	SE11	Puglia	ITF4		Unterfranken	DE26	11		Belgium
8	len	Swee	Campania	ITF3		Mittelfranken	DE25		Vorarlberg	AT34
	Canarias (ES)	ES70	Molise	ITF2		Oberfranken	DE24		Tirol	AT33
	Región de Murcia	ES62	Abruzzo	ITF1		Oberpfalz	DE23		Salzburg	AT 32
	Andalucia	ESG	Lazio	TTF-4		Viederhavern	DE39		Ohorösterreich	AT31
	Comunidad Valenciana Iller Bolone	ES52	Umbria	TTE2		Dharkanan	DE14		Stainmah	AT21
	Cataluña	ES51	Toscana	ITE1		Freiburg	DE13		Wien	AT13
	Extremadura	ES43	Emilia-Romagna	ITD5		Karlsruhe	DE12		Niederösterreich	AT12
	Castilla-la Mancha	ES42	Friuli-Venezia Giulia	ITD4	6	Stuttgart	DE11	¢	Burgenland	AT11
# regions	Costillo v Loón	BOD FOR	Veneto # re	D ITTD2	08 1012a1 #	IVAILLE	Cormany	0 stroffar #	DITIONT	Anstria
II wantawa	o Mana		N # 10	- I NITITE		North Anna Anna Anna Anna Anna Anna Anna Ann	INTITO	the maining	Warma	NITITO

## References

- ALTOMONTE, C., AND I. COLANTONE (2009): "Firm Heterogeneity and Endogenous Regional Disparities"," Journal of Economic Geography.
- AUDRETSCH, D. B., AND M. P. FELDMAN (2004): "Knowledge spillovers and the geography of innovation," in Handbook of Regional and Urban Economics, ed. by J. V. Henderson, and J. F. Thisse, vol. 4 of Handbook of Regional and Urban Economics, chap. 61, pp. 2713–2739. Elsevier.
- BARBA NAVARETTI, G., D. CASTELLANI, AND A.-C. DISDIER (2010): "How Does Investing in Cheap Labour Countries Affect Performance at Home? France and Italy," Oxford Economic Papers, doi:10.1093/oep/gpp010.

BARBA NAVARETTI, G., AND A. J. VENABLES (2006): Le multinazionali nell'economia mondiale. il Mulino.

- BASILE, R., D. CASTELLANI, AND A. ZANFEI (2008): "Location Choices of Multinational Firms in Europe: The Role of EU Cohesion Policy," *Journal of International Economics*, 74, 328–340.
- BITZER, J., AND H. GÖRG (2009): "Foreign direct investment, competition and industry performance," The World Economy, 31(2), 221–233.
- BLONIGEN, B. A., R. B. DAVIES, G. R. WADDELL, AND H. T. NAUGHTON (2007): "FDI in space: Spatial autoregressive relationships in foreign direct investment," *European Economic Review*, 51(5), 1303 – 1325.
- BLONIGEN, B. A., AND V. KOLPIN (2007): "Technology, agglomeration, and regional competition for investment.," Canadian Journal of Economics, 40(4), 1149 – 1167.
- BOBONIS, G. J., AND H. J. SHATZ (2007): "Agglomeration, Adjustment, and State Policies in the Location of Foreign Direct Investment in the United States," *The Review of Economics and Statistics*, 89(1), 30–43.
- BRACALENTE, B., AND C. PERUGINI (2008): "The components of regional disparities in Europe," The Annals of Regional Science.
- BRANSTETTER, L. (2006): "Is foreign direct investment a channel of knowledge spillovers? Evidence from Japan's FDI in the United States," Journal of International Economics, 68(2), 325 344.
- CANTWELL, J. A. (1995): "The Globalisation of Technology: What Remains of the Product Cycle Model?," Cambridge Journal of Economics, 19(1), 155–74.

CASTELLANI, D., AND A. ZANFEI (2006): Multinational Firms, Innovation and Productivity. Edward Elgar.

CICCONE, A. (2002): "Agglomeration effects in Europe," European Economic Review, 46(2), 213–227.

- CINGANO, F., AND F. SCHIVARDI (2004): "Identifying the Sources of Local Productivity Growth," Journal of the European Economic Association, 2(4), 720–742.
- CRESPO, N., M. P. FONTOURA, AND I. PROENÇA (2009): "FDI spillovers at regional level: Evidence from Portugal," Papers in Regional Science, 88(3), 591–607.
- CROZET, M., T. MAYER, AND J.-L. MUCCHIELLI (2004): "How do firms agglomerate? A study of FDI in France," Regional Science and Urban Economics, 34(1), 27–54.
- DAVIES, R. B. (2005): "State tax competition for foreign direct investment: a winnable war?," Journal of International Economics, 67(2), 498 512.
- DEBAERE, P., H. LEE, AND J. LEE (2010): "It Matters Where You Go. Outward Foreign Direct Investment and Multinationals' Employment Growth at Home," *Journal of Development Economics*, 91(2), 301–309.

DRIFFIELD, N. (2004): "Regional policy and spillovers from FDI in the UK," The Annals of Regional Science, 38, 4.

- DRIFFIELD, N., J. H. LOVE, AND K. TAYLOR (2009): "Productivity and labor demand effects of inward and outward FDI on UK industry," *The Manchester School*, 77, 171–203.
- ELHORST, J. P. (2010): "Spatial Panel Data Models," in *Handbook of Applied Spatial Analysis*, ed. by M. Fischer, and A. Getis. Springer, Berlin.

FELIX, B. (2006): "Statistics in Focus - Science and Technology," Discussion paper, Eurostat.

- FOSFURI, A., AND M. MOTTA (1999): "Multinationals without Advantages," Scandinavian Journal of Economics, 101(4), 617–30.
- GAMBARDELLA, A., M. MARIANI, AND S. TORRISI (2008): "How 'Provincial' is your Region? Openness and Regional Performance in Europe," *Regional Studies*, pp. 1–13.
- GIRMA, S., AND K. WAKELIN (2007): "Local productivity spillovers from foreign direct investment in the U.K. electronics industry," *Regional Science and Urban Economics*, 37(3), 399–412.
- Görg, H., AND E. STROBL (2005): "Foreign direct investment and local economic development: Beyond productivity spillovers," in *Does Foreign Direct Investment Promote Development?*, ed. by T. Moran, E. Graham, and M. Blomström. Institute for International Economics, Washington DC.
- GREENAWAY, D., AND R. KNELLER (2007): "Firm heterogeneity, exporting and foreign direct investment," *Economic Journal*, 117(517), F134–F161.
- GRIFFITH, R., R. HARRISON, AND J. V. REENEN (2006): "How Special Is the Special Relationship? Using the Impact of U.S. R&D Spillovers on U.K. Firms as a Test of Technology Sourcing," *American Economic Review*, 96(5), 1859–1875.
- GRILICHES, Z., AND J. MAIRESSE (1995): "Production Functions: The Search for Identification," NBER Working Papers 5067, National Bureau of Economic Research, Inc.
- HALL, B. H., AND J. MAIRESSE (1995): "Exploring the relationship between R&D and productivity in French manufacturing firms," *Journal of Econometrics*, 65, 263–293.
- HALPERN, L., AND B. MURAKOZY (2007): "Does distance matter in spillover?," The Economics of Transition, 15, 781-805.
- HEAD, K., AND T. MAYER (2004): "Market Potential and the Location of Japanese Investment in the European Union," The Review of Economics and Statistics, 86(4), 959–972.

HELPMAN, E., M. MELITZ, AND S. YEAPLE (2004): "Export versus FDI with heterogeneous firms," American Economic Review, 94(1), 300–316.

KRUGMAN, P. (1993): Geography and Trade. MIT Press.

- MILLO, G., AND G. PIRAS (2009): "splm: spatial panel data models in R," .
- MULLEN, J., AND M. WILLIAMS (2007): "FOREIGN DIRECT INVESTMENT AND REGIONAL PRODUCTIVITY SPILLOVERS IN US MANUFACTURING," Review of Urban & Regional Development Studies, 19(3), 185–196.
- NICOLETTI, G., AND S. SCARPETTA (2003): "Regulation, productivity and growth: OECD evidence," *Economic Policy*, 18(36), 9–72.
- O'MAHONY, M., A. RINCON-AZNAR, AND C. ROBINSON (2010): "Productivity Growth in Europe and the US: a Sectoral Study," Review of Economics and Institutions, 1(1).
- PACI, R., AND S. USAI (2000): "Technological Enclaves and Industrial Districts: An Analysis of the Regional Distribution of Innovative Activity in Europe," *Regional Studies*, 34(2), 97–114.
- PETIT, M.-L., AND F. SANNA-RANDACCIO (2000): "Endogenous R&D and foreign direct investment in international oligopolies," International Journal of Industrial Organization, 18(2), 339–367.
- SALZ, P., E. BUISMAN, J. SMIT, AND B. DE VOS (2006): "Employment in the fisheries sector: current situation," Final report, Eurostat.
- VAHTER, P., AND J. MASSO (2007): "Home versus Host Country Effects of FDI: Searching for New Evidence of Productivity Spillovers," Applied Economics Quarterly (formerly: Konjunkturpolitik), 53(2), 165–196.
- VAN POTTELSBERGHE DE LA POTTERIE, B., AND F. LICHTENBERG (2001): "Does Foreign Direct Investment Transfer Technology across Borders?," *Review of Economics and Statistics*, 82, 490–97.
- VENABLES, A. J. (1996): "Equilibrium Locations of Vertically Linked Industries," International Economic Review, 37(2), 341–59.

# QUADERNI DEL DIPARTIMENTO DI ECONOMIA, FINANZA E STATISTICA Università degli Studi di Perugia

1	Gennaio 2005	Giuseppe CALZONI Valentina BACCHETTINI	Il concetto di competitività tra approccio classico e teorie evolutive. Caratteristiche e aspetti della sua determinazione
2	Marzo 2005	Fabrizio LUCIANI Marilena MIRONIUC	Ambiental policies in Romania. Tendencies and perspectives
3	Aprile 2005	Mirella DAMIANI	Costi di agenzia e diritti di proprietà: una premessa al problema del governo societario
4	Aprile 2005	Mirella DAMIANI	Proprietà, accesso e controllo: nuovi sviluppi nella teoria dell'impresa ed implicazioni di corporate governance
5	Aprile 2005	Marcello SIGNORELLI	Employment and policies in Europe: a regional perspective
6	Maggio 2005	Cristiano PERUGINI Paolo POLINORI Marcello SIGNORELLI	An empirical analysis of employment and growth dynamics in the italian and polish regions
7	Maggio 2005	Cristiano PERUGINI Marcello SIGNORELLI	Employment differences, convergences and similarities in italian provinces
8	Maggio 2005	Marcello SIGNORELLI	Growth and employment: comparative performance, convergences and co- movements
9	Maggio 2005	Flavio ANGELINI Stefano HERZEL	Implied volatilities of caps: a gaussian approach
10	Giugno 2005	Slawomir BUKOWSKI	EMU – Fiscal challenges: conclusions for the new EU members
11	Giugno 2005	Luca PIERONI Matteo RICCIARELLI	Modelling dynamic storage function in commodity markets: theory and evidence
12	Giugno 2005	Luca PIERONI Fabrizio POMPEI	Innovations and labour market institutions: an empirical analysis of the Italian case in the middle 90's
13	Giugno 2005	David ARISTEI Luca PIERONI	Estimating the role of government expenditure in long-run consumption
14	Giugno 2005	Luca PIERONI Fabrizio POMPEI	Investimenti diretti esteri e innovazione in Umbria
15	Giugno 2005	Carlo Andrea BOLLINO Paolo POLINORI	Il valore aggiunto su scala comunale: la Regione Umbria 2001-2003
16	Giugno 2005	Carlo Andrea BOLLINO Paolo POLINORI	Gli incentivi agli investimenti: un'analisi dell'efficienza industriale su scala geografica regionale e sub regionale

17	Giugno 2005	Antonella FINIZIA Riccardo MAGNANI Federico PERALI Paolo POLINORI Cristina SALVIONI	Construction and simulation of the general economic equilibrium model Meg-Ismea for the italian economy
18	Agosto 2005	Elżbieta KOMOSA	Problems of financing small and medium-sized enterprises. Selected methods of financing innovative ventures
19	Settembre 2005	Barbara MROCZKOWSKA	Regional policy of supporting small and medium-sized businesses
20	Ottobre 2005	Luca SCRUCCA	Clustering multivariate spatial data based on local measures of spatial autocorrelation
21	Febbraio 2006	Marco BOCCACCIO	Crisi del welfare e nuove proposte: il caso dell'unconditional basic income
22	Settembre 2006	Mirko ABBRITTI Andrea BOITANI Mirella DAMIANI	Unemployment, inflation and monetary policy in a dynamic New Keynesian model with hiring costs
23	Settembre 2006	Luca SCRUCCA	Subset selection in dimension reduction methods
24	Ottobre 2006	Sławomir I. BUKOWSKI	The Maastricht convergence criteria and economic growth in the EMU
25	Ottobre 2006	Jan L. BEDNARCZYK	The concept of neutral inflation and its application to the EU economic growth analyses
26	Dicembre 2006	Fabrizio LUCIANI	Sinossi dell'approccio teorico alle problematiche ambientali in campo agricolo e naturalistico; il progetto di ricerca nazionale F.I.S.R. – M.I.C.E.N.A.
27	Dicembre 2006	Elvira LUSSANA	Mediterraneo: una storia incompleta
28	Marzo 2007	Luca PIERONI Fabrizio POMPEI	Evaluating innovation and labour market relationships: the case of Italy
29	Marzo 2007	David ARISTEI Luca PIERONI	A double-hurdle approach to modelling tobacco consumption in Italy
30	Aprile 2007	David ARISTEI Federico PERALI Luca PIERONI	Cohort, age and time effects in alcohol consumption by Italian households: a double-hurdle approach
31	Luglio 2007	Roberto BASILE	Productivity polarization across regions in Europe
32	Luglio 2007	Roberto BASILE Davide CASTELLANI Antonello ZANFEI	Location choices of multinational firms in Europe: the role of EU cohesion policy
33	Agosto 2007	Flavio ANGELINI Stefano HERZEL	Measuring the error of dynamic hedging: a Laplace transform approach

3/	Agosto 2007	Stefano HERZEI	The ICARCH effect: consequences on
J <del>1</del>	ngosto 2007	Cătălia STĂDICĂ	volatility foregasting and option
			trading
25	<b>A A O</b> 007		
35	Agosto 2007	Flavio ANGELINI	Explicit formulas for the minimal
		Stefano HERZEL	variance hedging strategy in a
			martingale case
36	Agosto 2007	Giovanni BIGAZZI	The role of agriculture in the
			development of the people's Republic
			of China
37	Settembre 2007	Enrico MARELLI	Institutional change, regional features
		Marcello SIGNORELLI	and aggregate performance in eight
			EU's transition countries
38	Ottobre 2007	Paolo NATICCHIONI	Wage structure, inequality and skill-
		Andrea RICCI	biased change: is Italy an outlier?
		Emiliano BUSTICHELLI	Shoed enhanger to rully un outlet.
30	Novembre 2007	The International Study	Exports and productivity. Comparable
57		Group on Exports and	evidence for 14 countries
		Productivity	evidence for 14 countries
40	Dicombro 2007	Castano MARTINO	Contracting food safety strategies in
40	Dicembre 2007	De ala DOLINORI	Contracting food safety strategies in
		Paolo POLINORI	hybrid governance structures
41	Dicembre 2007	Floro Ernesto CAROLEO	The youth experience gap:
		Francesco PASTORE	explaining differences across EU
			countries
42	Gennaio 2008	Melisso BOSCHI	Aluminium market and the
		Luca PIERONI	macroeconomy
43	Febbraio 2008	Elavio ANGELINI	Hedging error in Lévy models with a
15	1 0001210 2000	Marco NICOLOSI	fast Fourier Transform approach
44	Febbraio 2008	Luca PIERONI	Can we declare military Keynesianism
		Giorgio d'AGOSTINO	dead?
		Marco LORUSSO	
45	Febbraio 2008	Pierluigi GRASSELLI	Mediterranean models of Welfare
		Cristina MONTESI	towards families and women
		Paola IANNONE	
46	Marzo 2008	Mirella DAMIANI	Mergers, acquisitions and
		Fabrizio POMPEI	technological regimes: the European
			experience over the period 2002-2005
47	Marzo 2008	Bruno BRACALENTE	The Components of Regional
			D'analitica in Economi
		Cristiano PERUGINI	Disparities in Europe
48	Marzo 2008	Cristiano PERUGINI Cristiano PERUGINI	FDL R&D and Human Capital in
48	Marzo 2008	Cristiano PERUGINI Cristiano PERUGINI Eabrizio POMPEI	FDI, R&D and Human Capital in Central and Eastern European
48	Marzo 2008	Cristiano PERUGINI Cristiano PERUGINI Fabrizio POMPEI Marcello SIGNORELLI	FDI, R&D and Human Capital in Central and Eastern European
48	Marzo 2008	Cristiano PERUGINI Cristiano PERUGINI Fabrizio POMPEI Marcello SIGNORELLI Cristiano PERUCINI	FDI, R&D and Human Capital in Central and Eastern European Countries
48 49	Marzo 2008 Marzo 2008	Cristiano PERUGINI Cristiano PERUGINI Fabrizio POMPEI Marcello SIGNORELLI Cristiano PERUGINI	FDI, R&D and Human Capital in Central and Eastern European Countries Employment and Unemployment in the Italian Provinces
48	Marzo 2008 Marzo 2008	Cristiano PERUGINI Cristiano PERUGINI Fabrizio POMPEI Marcello SIGNORELLI Cristiano PERUGINI	FDI, R&D and Human Capital in Central and Eastern European Countries Employment and Unemployment in the Italian Provinces
48 49 50	Marzo 2008 Marzo 2008 Marzo 2008	Cristiano PERUGINI Cristiano PERUGINI Fabrizio POMPEI Marcello SIGNORELLI Cristiano PERUGINI Sławomir I. BUKOWSKI	Disparities in EuropeFDI, R&D and Human Capital in Central and Eastern European CountriesEmployment and Unemployment in the Italian ProvincesOn the road to the euro zone.
48 49 50	Marzo 2008 Marzo 2008 Marzo 2008	Cristiano PERUGINI Cristiano PERUGINI Fabrizio POMPEI Marcello SIGNORELLI Cristiano PERUGINI Sławomir I. BUKOWSKI	Disparities in EuropeFDI, R&D and Human Capital in Central and Eastern European CountriesEmployment and Unemployment in the Italian ProvincesOn the road to the euro zone. Currency rate stabilization: experiences
48 49 50	Marzo 2008 Marzo 2008 Marzo 2008	Cristiano PERUGINI Cristiano PERUGINI Fabrizio POMPEI Marcello SIGNORELLI Cristiano PERUGINI Sławomir I. BUKOWSKI	FDI, R&D and Human Capital in Central and Eastern European Countries Employment and Unemployment in the Italian Provinces On the road to the euro zone. Currency rate stabilization: experiences of the selected EU countries
48 49 50 51	Marzo 2008 Marzo 2008 Marzo 2008 Aprile 2008	Cristiano PERUGINI Cristiano PERUGINI Fabrizio POMPEI Marcello SIGNORELLI Cristiano PERUGINI Sławomir I. BUKOWSKI Bruno BRACALENTE	Disparities in EuropeFDI, R&D and Human Capital in Central and Eastern European CountriesEmployment and Unemployment in the Italian ProvincesOn the road to the euro zone. Currency rate stabilization: experiences of the selected EU countriesHomogeneous, Urban Heterogeneous,
48 49 50 51	Marzo 2008 Marzo 2008 Marzo 2008 Aprile 2008	Cristiano PERUGINI Cristiano PERUGINI Fabrizio POMPEI Marcello SIGNORELLI Cristiano PERUGINI Sławomir I. BUKOWSKI Bruno BRACALENTE Cristiano PERUGINI	Disparities in EuropeFDI, R&D and Human Capital in Central and Eastern European CountriesEmployment and Unemployment in the Italian ProvincesOn the road to the euro zone. Currency rate stabilization: experiences of the selected EU countriesHomogeneous, Urban Heterogeneous, or both? External Economies and
48 49 50 51	Marzo 2008 Marzo 2008 Marzo 2008 Aprile 2008	Cristiano PERUGINI Cristiano PERUGINI Fabrizio POMPEI Marcello SIGNORELLI Cristiano PERUGINI Sławomir I. BUKOWSKI Bruno BRACALENTE Cristiano PERUGINI Fabrizio POMPEI	Disparities in EuropeFDI, R&D and Human Capital in Central and Eastern European CountriesEmployment and Unemployment in the Italian ProvincesOn the road to the euro zone. Currency rate stabilization: experiences of the selected EU countriesHomogeneous, Urban Heterogeneous, or both? External Economies and Regional Manufacturing Productivity
48 49 50 51	Marzo 2008 Marzo 2008 Marzo 2008 Aprile 2008	Cristiano PERUGINI Cristiano PERUGINI Fabrizio POMPEI Marcello SIGNORELLI Cristiano PERUGINI Sławomir I. BUKOWSKI Bruno BRACALENTE Cristiano PERUGINI Fabrizio POMPEI	Disparities in EuropeFDI, R&D and Human Capital in Central and Eastern European CountriesEmployment and Unemployment in the Italian ProvincesOn the road to the euro zone. Currency rate stabilization: experiences of the selected EU countriesHomogeneous, Urban Heterogeneous, or both? External Economies and Regional Manufacturing Productivity in Europe

52	Aprile 2008	Gaetano MARTINO	Income inequality within European
52	11pme 2000	Cristiano PERUGINI	regions: determinants and effects on growth
53	Aprile 2008	Ian L. BEDNARCZYK	Controversy over the interest rate
00	11pme 2000		theory and policy. Classical approach
			to interest rate and its continuations
54	Aprile 2008	Bruno BRACALENTE	Factor decomposition of cross-
• •	p	Cristiano PERUGINI	country income inequality with
			interaction effects
55	Aprile 2008	Cristiano PERUGINI	Employment Intensity of Growth in
	1		Italy. A Note Using Regional Data
56	Aprile 2008	Cristiano PERUGINI	Technological Change, Labour
	Γ	Fabrizio POMPEI	Demand and Income Distribution in
			European Union Countries
57	Aprile 2008	Simona BIGERNA	L'analisi delle determinanti della
	1	Paolo POLINORI	domanda di trasporto pubblico nella
			città di Perugia
58	Maggio 2008	Simona BIGERNA	The willingness to pay for Renewable
		Paolo POLINORI	Energy Sources (RES): the case of
			Italy with different survey approaches
			and under different EU "climate
			vision". First results
59	Giugno 2008	Simona BIGERNA	Ambiente operativo ed efficienza nel
		Paolo POLINORI	settore del Trasporto Pubblico Locale
			in Italia
60	Ottobre 2008	Pierluigi GRASSELLI	L'interpretazione dello spirito del
		Cristina MONTESI	dono
(1		Roberto VIRDI	TY 1. 1 1 .
61	Novembre 2008	Antonio BOGGIA	L'impatto ambientale ed economico
		Fabrizio LUCIANI	del cambiamento climatico
		Giannuca MASSEI	suil agricoltura
62	Novembre 2008	Elepa STANCHELLINI	On the estimation of a binary response
02	NOVEINDLE 2008	Erancesco Claudio STINGO	model in a selected population
		Rosa CAPOBIANCO	model in a selected population
63	Dicembre 2008	Gianna FIGÀ-TALAMANCA	Limit results for discretely observed
00	Dicembre 2000		stochastic volatility models with
			leverage effect
64	Maggio 2009	Mirella DAMIANI	Factors behind performance-related
		Andrea RICCI	pay: evidence from Italy
65	Giueno 2009	Alessandra RIGHI	The Timing of the School-to-
	01005110 2007	Dario SCIULLI	Permanent Work Transition: a
			Comparison across Ten European
			Countries
66	Settembre 2009	Fabrizio LUCIANI	Economia agraria e pianificazione
			economica territoriale nel Parco
			nazionale del Sagarmatha (Everest,
			Nepal)
67	Settembre 2009	Valentina TIECCO	I regimi di protezione dell'impiego

68	Ottobre 2009	Gianna FIGÀ-TALAMANCA	Path properties of simulation schemes for the Heston stochastic volatility
			model
69	Ottobre 2009	Cristina MONTESI	A comparative analysis of different business ethics in the perspective of the Common Good
70	Ottobre 2009	Luisa FRANZINI	Determinants of Health Disparities in
		Margherita GIANNONI	Italian Regions
71	Novembre 2009	Flavio ANGELINI	Evaluating Discrete Dynamic
		Stefano HERZEL	Strategies in Affine Models
72	Novembre 2009	Giuseppe ARBIA Michele BATTISTI	Institutions and geography: Empirical test of spatial growth models for
		Gianfranco DI VAIO	European regions
73	Gennaio 2010	Mirella DAMIANI Andrea RICCI	Performance-Related Pay, Unions and Productivity in Italy: evidence from quantile regressions
74	Febbraio 2010	Davide CASTELLANI Fabio PIERI	The Effect of Foireign Investments on European Regional Productivity
75	Luglio 2010	Guglielmo M. CAPORALE Davide CIFERRI Alessandro GIRARDI	Time-varying spot and futures oil price dynamics
76	Settembre 2010	Mirella DAMIANI	Labour regulation, corporate governance and varieties of capitalism
77	Settembre 2010	Dario SCIULLI Marcello SIGNORELLI	University-to-work transitions: the case of Perugia
78	Ottobre 2010	Olga DEMIDOVA Marcello SIGNORELLI	The Impact of Crises on Youth Unemployment of Russian Regions: An Empirical Analysis
79	Ottobre 2010	Misbah T. CHOUDHRY Enrico MARELLI Marcello SIGNORELLI	The Impact of Financial Crises on Youth Unemployment Rate
80	Novembre 2010	Marco BELLUCCI	Fusioni ed acquisizioni: determinanti ed effetti in un confronto europeo
81	Dicembre 2010	Silvia MICHELI	Learning Curve and Wind Power
82	Dicembre 2010	Leonardo BECCHETTI Stefano CASTRIOTA Elena GIACHIN RICCA	Beyond the Joneses: inter-country income comparisons and happiness
83	Gennaio 2011	Davide CASTELLANI Fabio PIERI	Foreign Investments and Productivity Evidence from European Regions

# I QUADERNI DEL DIPARTIMENTO DI ECONOMIA Università degli Studi di Perugia

1	Dicembre 2002	Luca PIERONI:	Further evidence of dynamic demand systems in three european
			countries
2	Dicembre 2002	Luca PIERONI	Il valore economico del paesaggio:
		Paolo POLINORI:	un'indagine microeconomica
3	Dicembre 2002	Luca PIERONI	A note on internal rate of return
		Paolo POLINORI:	
4	Marzo 2004	Sara BIAGINI:	A new class of strategies and
-			application to utility maximization
			for unbounded processes
5	Aprile 2004	Cristiano PERUGINI:	La dipendenza dell'agricoltura
J	ripine 2001		italiana dal sostegno pubblico:
			un'analisi a livello regionale
6	Maggio 2004	Mirollo DAMIANI.	Nuova macroaconomia kovnosiana
0	Maggio 2004		e cuosi razionalità
7	Maggio 2004	Mauro VISACCIO:	Dimensione e persistenza degli
1	Maggio 2004	Wadio v 13/10010.	aggiustamonti fiscoli in prosonza di
			debito pubblico elevato
	Marcia 2004	Maura VISACCIO:	Door the growth stability past
0	Magg10 2004	Mauto VISAGGIO.	Does the growth stability pact
			Freed with a dequate and consistent
	0. 0004		
9	Giugno 2004		Kedistribution and labour market
	<u></u>	Francesco FARINA:	institutions in OECD countries
10	Giugno 2004	Marco BOCCACCIO:	Tra regolamentazione settoriale e
			antitrust: il caso delle
	<u></u>		telecomunicazioni
11	Giugno 2004	Cristiano PERUGINI	Labour market performance in
		Marcello SIGNORELLI:	central european countries
12	Luglio 2004	Cristiano PERUGINI	Labour market structure in the
		Marcello SIGNORELLI:	italian provinces: a cluster analysis
13	Luglio 2004	Cristiano PERUGINI	I flussi in entrata nei mercati del
		Marcello SIGNORELLI:	lavoro umbri: un'analisi di cluster
14	Ottobre 2004	Cristiano PERUGINI:	Una valutazione a livello
			microeconomico del sostegno
			pubblico di breve periodo
			all'agricoltura. Il caso dell'Umbria
			attraverso i dati RICA-INEA
15	Novembre 2004	Gaetano MARTINO	Economic inequality and rural
		Cristiano PERUGINI	systems: empirical evidence and
			interpretative attempts
16	Dicembre 2004	Federico PERALI	Bilancio ambientale delle imprese
		Paolo POLINORI	agricole italiane: stima
		Cristina SALVIONI	dell'inquinamento effettivo
		Nicola TOMMASI	•
		Marcella VERONESI	